

INTER-ORGANISATIONAL IT GOVERNANCE

- A Case Study of Municipal ICT Cooperation

Ari Helin



INTER-ORGANISATIONAL IT GOVERNANCE

- A Case Study of Municipal ICT Cooperation

Ari Helin

University of Turku

Turku School of Economics Department of Management and Entrepreneurship Information Systems Science Doctoral programme of Turku School of Economics

Supervised by

Professor, Jukka Heikkilä University of Turku Professor of Practice, Tomi Dahlberg University of Turku

Reviewed by

Professor, Tero Päivärinta Luleå University of Technology Assistant Professor, Anant S. Joshi Maastricht University

Opponent

Professor, Tero Päivärinta Luleå University of Technology

The originality of this publication has been checked in accordance with the University of Turku quality assurance system using the Turnitin OriginalityCheck service.

ISBN 978-951-29-8088-8 (PRINT) ISBN 978-951-29-8089-5 (PDF) ISSN 2343-3159 (Print) ISSN 2343-3167 (Online) Painosalama Oy, Turku, Finland 2020



UNIVERSITY OF TURKU

Turku School of Economics

Department of Management and Entrepreneurship

Information Systems Science

ARI HELIN: Inter-Organisational IT Governance – A Case Study of Municipal

ICT Cooperation

Doctoral Dissertation, 147 pp.

Doctoral Programme of Turku School of Economics

August 2020

ABSTRACT

Changes in society and organisations' business environments have increased the use of operating models, which favour inter-organisational cooperation and networking. The evolution of operating models also has a significant impact on organisations' IT departments and ICT activities. Inter-organisational ICT cooperation and networking offer excellent opportunities to public organisations to develop and improve their operations.

This study clarifies how public organisations in Finland have implemented interorganisational ICT cooperation. Especially the inter-organisational ICT cooperation between Finnish municipalities has been the target of the current study. The study clarifies why municipalities plan ICT cooperation and how they have implemented it. The study also reviews the benefits that municipalities have received or are seeking to obtain through ICT cooperation. In addition, the study clarifies how intermunicipal IT governance has been implemented and investigates the applicability of currently used theories explaining intra-organisational IT governance to describe inter-organisational IT governance.

The study examines various inter-municipal ICT cooperation groups in Finland. The study reveals that even though municipality groups use many types of ICT cooperation, joint IT governance model-based decisions and clear goals often shine in their absence. The dominant form of collaboration in inter-municipal ICT cooperation is voluntary cooperation without any joint IT governance. The study reveals that without a joint IT governance, municipalities lose much of the benefits they could achieve through a joint IT governance model.

The study also confirms the applicability of currently used theories describing the intra-organisational IT governance to describe inter-organisational IT governance.

KEYWORDS: information technology, IT, IT governance, ITG, information and communication technology, ICT, inter-organisational, public organisations, municipal, ICT cooperation

TURUN YLIOPISTO

Kauppakorkeakoulu

Johtamisen ja yrittäjyyden laitos

Tietojärjestelmätiede

ARI HELIN: Organisaatioiden välinen IT-hallinto - tapaustutkimus kuntien

ICT-yhteistyöstä

Väitöskirja, 147 s.

Turun kauppakorkeakoulun tohtoriohjelma

elokuu 2020

TIIVISTELMÄ

Muutokset yhteiskunnassa ja organisaatioiden toimintaympäristöissä ovat lisänneet sellaisten toimintamallien käyttöä, jotka suosivat organisaatioiden yhteistyötä ja verkottumista. Näillä muutoksilla on myös voimakas vaikutus organisaatioiden IT-osastoihin ja ICT-toimintoihin. Uudenlaiset yhteistyön toimintamallit ovat luoneet tarpeen myös eri organisaatioiden yhteiselle IT-hallinnolle. Myös julkishallinnon organisaatioille tällainen yhteistyö tarjoaa hyvät mahdollisuudet kehittää ja tehostaa toimintaansa.

Tässä tutkimuksessa selvitetään, kuinka Suomen julkishallinnossa toteutetaan organisaatioiden välistä ICT-yhteistyötä. Tutkimuskohteena on erityisesti kuntien välinen ICT-yhteistyö. Tutkimuksessa selvitetään, miksi kunnat suunnittelevat ICT-yhteistyötä, miten ne toteuttavat sitä sekä miten tätä yhteistyötä hallitaan. Tutkimuksessa tarkastellaan myös niitä hyötyjä, joita kunnat ovat ICT-yhteistyön avulla saaneet tai pyrkineet saamaan. Lisäksi selvitetään, voidaanko nykyisin käytössä olevia, organisaatioiden sisäistä IT-hallintoa kuvaavia teorioita käyttää kuvaamaan organisaatioiden välistä IT-hallintoa.

Tutkimuksessa on kartoitettu suomalaisten kuntien välisiä ICT-yhteistyöryhmiä. Tutkimuksessa selviää, että vaikka kuntaryhmät tekevät monenlaista ICT-yhteistyötä, niin yhteiseen IT-hallintomalliin perustuvat päätökset ja yhteiset, selkeät tavoitteet loistavat usein poissaolollaan. Kuntien välisessä ICT-yhteistyössä hallitseva yhteistyömalli on vapaaehtoinen yhteistyö ilman yhteistä IT-hallintomallia. Tutkimuksen perusteella voi todeta, että kunnat menettävät suuren osan niistä hyödyistä, jotka ne voisivat saavuttaa yhteisen IT-hallintomallin avulla.

Tutkimus vahvistaa nykyisin käytössä olevien, organisaatioiden sisäistä IT-hallintoa kuvaavien teorioiden käyttökelpoisuuden kuvaamaan organisaatioiden välistä IT-hallintoa.

ASIASANAT: tietotekniikka, IT, IT-hallinto, tieto- ja viestintätekniikka, ICT, organisaatioiden välinen, julkinen sektori, kunta, ICT-yhteistyö

Acknowledgemets

This dissertation project has been undoubtedly a surprise to many and not least to myself. My work career continued already for 30 years, full of international business speed and challenging situations before the academic world took me with it. Challenges in the academic world are different compared to the business environment, but I found the academic world very positive and refreshing indeed. Possibility to do something different and learn new things drove me to doctoral studies. A big thank you to D.Sc. Jussi Puhakainen for starting the dissertation project.

My original idea was that the doctoral thesis had to be a continuum to my master thesis. However, changes in the situations in my working life and a work project in the university directed me eventually to change the topic of the thesis to a new one, which proved to be extremely interesting and topical one. Many thanks for the continuation of the dissertation project belong to my supervisor, Professor Jukka (Jups) Heikkilä, who took me along to an inter-municipal IT project. This new environment became a central part of my learning and my dissertation project. Big thanks to all not named IT experts, whose work I have been allowed to follow in the project and who I interviewed for my doctoral thesis. The recognition also belongs to Professor of Practice Tomi Dahlberg from whom I learned the skills of writing academic articles when we produced the articles for my dissertation together. Likewise, my thanks belong to Professor Hannu Salmela, who patiently followed and guided my studies from the B.Sc. to D.Sc.

I am incredibly grateful to the reviewers of my thesis, Professor Tero Päivärinta and Assistant Professor Anant S. Joshi. They had the strength to go into my thesis and gave wise instructions and valuable comments to develop it. At the same time big thanks to Professor Tero Päivärinta from acting as my opponent.

During my postgraduate studies, I have been allowed to become acquainted with many great people from whom I have learned life skills and philosophies and who have supported me in different challenges of my dissertation project. The special thanks belong to Lecturer Antti Tuomisto who has given me a place for work in his continually changing premises, who has taken me along to the challenging work projects and who has acted as an example of dedicated teacher and leader. The big

thanks also belong to Associate Professor Sami Hyrynsalmi who has had time to give me good advice and guidance despite his fast-rising career ladder.

The warm thanks also belong to all my friends and colleagues whom I have been lucky to share ideas during my doctoral studies. Thanks to Ph.D. Jani Koskinen, whose extensive ethical knowledge and skills of the traditional building I have absorbed and Professor of Practice Janne Lahtiranta, whose philosophy of the life I was able to enjoy. Thanks to my student colleagues Minna Rantala and Tiina Nokkala with whom I shared sorrows and challenges of the dissertation project. Furthermore, I thank Adjunct Professor Kai Kimppa, whose ethical understanding of life is admirable, and my colleagues D.Sc. Ville Harkke and Eva Collanus for the praiseworthy cooperation in our work project. Thanks also to my other colleagues, Nico Arvela, Sari Knaapi-Junnila, Juhani Naskali, Mikko Vermanen and Jesse Kaukola. They made even the grey working days genial and worth of living. My humble thanks to Birgit Haanpää, who helped me in practical challenges in the early stages of my project. Thanks also to those numerous not named members of the staff of Turku School of Economics about positive cooperation and friendliness.

In addition to the Department of the Information Systems Science of the Turku School of Economics, I want to thank the Jenny and Antti Wihuri Foundation, the Turku School of Economics Support Foundation and Princeps Oy as economic supporters of my dissertation project.

Above all, I own my family the biggest thanks. My wife Kirsti showed me the way to lifelong studies, and she supported me in my moments of despair. My children Henrik and Katariina challenged me through my doctoral studies and coached me to keep up with younger generations. Without your existence, I would not write these words either.

24.6.2020 *Ari Helin*

Table of Contents

Ac	knowl	edgemets	6
Та	ble of	Contents	8
Ex	planat	ions of Terms and Phrases	10
Lis	t of O	riginal Publications	11
1	Intro	duction	13
	1.1	Background of the Research	13
	1.2	Motivation and Research Gap	16
	1.3	Research Objectives and RQs	17
	1.4	Overview of the Data Collection and Articles	
		1.4.1 Data sources, data collection and data processing	19
	1.5	1.4.2 The articles included in the thesis	22
	1.5	The Structure of the Thesis	23
2	Thec	retical Background	24
_	2.1	IT Governance	24
	2.2	Inter-organisational Cooperation Framework	26
		2.2.1 ŘBV theory	27
		2.2.2 TCE theory	29
	0.0	2.2.3 Social network theory	30
	2.3	Resource Dependency Theory	32
3	Posc	earch Approach and Procedure	33
J	3.1	Methodology	33
	3.2	Overview to Research Methods and Research Processes	35
	3.3	Author's Role in the Articles	39
4		irical Studies and Results	40
	4.1	The Articles	
		4.1.1 Article I	
		4.1.2 Article II	
		4.1.3 Article III	
		4.1.4 Article IV	
	4.2	Results	

5 Impl	lications and Conclusions	49
5.1	Implications	49
5.2	Theoretical Contribution	50
5.3	Practical Contributions	51
5.4	Limitations	52
5.5	Future Research	53
5.6	Conclusions	54
List of R	References	56
Append	ix	64
Original	Publications	67
J		
List of ta	ables	
Table 1 Table 2	Interviews and IntervieweesResearch Questions and Related Articles	20 21

Explanations of Terms and Phrases

This section clarifies some of the terms and phrases used in the thesis to prevent confusion.

- We have used the terms 'IT' and 'ICT' interchangeably. International Standards Organization (ISO) defines information technology (IT) as 'resources used to acquire, process, store and disseminate information' (ISO/IEC 38500, 2015). Communication technology (CT) is included in forming information and communications technology (ICT); thus, the terms IT and ICT have been used in this dissertation as synonyms depending on the commonly used terms.
- The term 'public sector' refers to the government and the local authority activities including, e.g., the healthcare services, social services, local public services and companies owned by the government (Oxford Reference, 2019). Similarly, 'public services' are services offered by the public sector.
- We have used the terms 'intra-organisational' and 'inter-organisational' widely in this dissertation. The term 'intra-organisational' describes issues and relationships which are taking place inside an organisation while the term 'inter-organisational' describes issues and relationships taking place between organisations.
- With the term 'IT governance', we mean various structures and procedures of decision-making. The term 'IT governance' includes the sharing of decision-making rights to people and organisations as well as decisions about the topics, which have to be decided. Also, 'IT governance' includes how the decisions will be made and who has the rights to make decisions. 'IT governance' is associated with the leadership of IT use (Zarvic et al. 2012; Weill & Ross, 2004). 'IT governance' must not be confused with 'IT management', which focuses on managing IT-related assets and issues on a day-to-day basis and ensuring that governance instructions and decisions are followed. 'IT governance' is much broader than 'IT management', but they are strongly dependent on each other (Pereira & da Silva, 2012). Since municipal ICT cooperation is very much based on agreements between municipalities about joint targets, structures and decision-making, this thesis focuses on 'IT governance'.

List of Original Publications

This dissertation is based on the following original publications, which are referred to in the text by their Roman numerals:

- I Helin, A. Inter-organizational IT governance research: A literature review. *International Journal of IT/Business Alignment and Governance (IJITBAG)*, 2019; 10(1), doi:10.4018/IJITBAG.2019010103.
- II Dahlberg, T., & Helin, A. Formation of voluntary inter-organizational IT governance for healthcare and social welfare IT Theoretical background and empirical evaluation. *Proceedings of the 8th International Conference on Theory and Practice of Electronic Governance (ICEGOV2014), Guimarães, Portugal*, October, 2014; doi:10.1145/2691195.2691230: 320-329.
- III Helin, A., & Dahlberg, T. Volume, benefits and factors that influence intermunicipal ICT cooperation in relation to ICT-related social services and healthcare services. *Finnish Journal of eHealth and eWelfare (FinJeHeW)*, 2017; 9(4), 299-312. doi:10.23996/fjhw.61065.
- IV Dahlberg, T., & Helin, A. How IT governance practices contribute to intermunicipal ICT cooperation and its benefits: Indeed, "the emperor has no clothes". *International Journal of IT/Business Alignment and Governance* (*IJITBAG*), 2017; 8(2), 62-79. doi:10.4018/IJITBAG.
- V Helin, A. SECOs as a mean to survive? Case municipal ICT. *Proceedings of the 9th International Workshop on Software Ecosystem (IWSECO2017), Espoo, Finland;* November, 2017; Retrieved from http://ceur-ws.org/Vol-2053/.

The original publications have been reproduced with the permission of the copyright holders.

The author of the thesis has also been involved in the following publications which have discussed municipal ICT cooperation:

- Helin, A. Inter-organizational IT Co-operation between municipalities: A literature review. *Proceedings of the 38th Information Systems Research Conference in Scandinavia (IRIS 38)*, Oulu, Finland, August 2015.
- Dahlberg, T. & Helin, A. The emperor with no clothes Inter-organizational ICT cooperation within municipal regions. *Proceedings of the 6th International Conference Well-being in the Information Society (WIS2016)*, Tampere, Finland, September 2016.
- Dahlberg, T., & Helin, A. Why and how do municipal areas govern inter-organizational ICT cooperation: Indeed, "The emperor has no clothes". *Proceedings of the 25th European Conference on Information Systems (ECIS2017)*, Guimarães, Portugal, June 2017.

1 Introduction

This chapter presents the background of the thesis, declares the motivation and the research gap and presents the research objectives with the research questions (RQs). The chapter also explains the research process, introduces the articles included in the thesis and presents the structure of the thesis.

1.1 Background of the Research

The role of information technology (IT) is crucial in the business and public environment and the status of IT has been widely recognised (e.g. Weill & Ross, 2005; Ali & Green, 2008; De Haes & Van Grembergen, 2009). No companies or public organisations can manage without IT and ICT services, i.e. an organisation's success depends on its IT systems (Sohal & Fitzpatrick, 2002; Ali & Green, 2007; Amali et al., 2014). To be able to use IT systems optimally, IT has to be governed reasonably and professionally and to benefit from their IT investments, organisations attempt to make effective use of IT governance to manage and develop their systems with care (Weill & Ross, 2004; Prasad et al., 2010).

Effective IT governance also improves an organisation's profits (Weill & Ross, 2004), which attracts management interest. Being essential for an organisation's success, IT governance has stabilised its position during the last decades, and it is considered a critical success factor for organisations (Ali & Green, 2007; Lunardi et al., 2014; Zhang et al., 2016). IT governance has been studied mostly from an enterprise perspective; however, the network perspective has received less attention (Trang et al., 2013). The current thesis thus aims to offer a broader view of the topic.

While the role of ICT is becoming increasingly crucial for business organisations (Henderson & Venkatraman, 1999), it is considered essential for municipalities because most of their services are operated using ICT (Flak et al., 2009). In the country of the research, Finland, there were 311 municipalities in 2017 (Ministry of Finance, 2018a). Local authorities in the municipalities are responsible for providing essential services including basic ICT-related services for their residents (The Association of

Finnish Local and Regional Authorities, 2018). According to the Constitution of Finland (Ministry of Justice, 1999), the administration of municipalities is based on the self-government of their residents. This principle includes tasks and issues related to various municipal ICT services. Simultaneously, according to the Constitution (Ministry of Justice, 1999, p. 24),

a public administrative task may be delegated to others than public authorities only by an Act or by virtue of an Act, if this is necessary for the appropriate performance of the task and if basic rights and liberties, legal remedies and other requirements of good governance are not endangered. However, a task involving the significant exercise of public powers can only be delegated to public authorities.

Based on this statement, municipalities are not allowed to outsource critical services to private enterprises; however, cooperation with other municipalities is possible instead. ICT cooperation between municipalities means sharing some ICT related tasks or services and also sharing corresponding costs and resources between municipalities in question. With the help of ICT cooperation, municipalities strive to cut ICT related costs and improve services. With ICT cooperation, municipalities need some sort of joint IT governance to take care of joint efforts.

There are different sizes of municipalities in Finland. The smallest municipal in Finland has less than 100 residents, while the largest has over 635,000 residents (Population Register Centre Finland, 2018). Due to the different municipal sizes, municipal budgets and resources are different, which affects the services municipalities can provide, including ICT-related services, which differ significantly. While bigger municipalities can maintain ICT departments with tens or even hundreds of people to serve their residents, smaller municipalities sometimes have less than one person to take care of ICT-related services municipal-wide (Ministry of Finance, 2015). It may also seem like several municipalities have not realised the importance of ICT for their current and future success. Still, as municipalities provide most of their services using ICT, they cannot function without ICT.

The need for better municipal ICT is growing due to the long-planned large-scale reform (Ministry of Finance, 2018b) and the need for digitalisation of services. Currently, in the public administration, Finland has been divided into four levels in accordance with the EU NUTS 2016 classification (EUR-Lex, 2003). According to the abovementioned plan, social welfare, healthcare and regional government reform would change the public administration structure and the responsibilities of the different administration levels. While currently, the municipalities are responsible for providing health services in their area (Ministry of Finance, 2019a), the planned reform would transfer responsibilities for all social welfare and healthcare issues to the

regional level (Ministry of Finance, 2019b). According to the proposed reform, more than 50% of the municipal ICT resources, financial and human, would be transferred to the regional level, thus creating a new ICT environment for municipalities with fewer resources and fewer responsibilities. However, the plan is on hold, awaiting decisions by the new government. Nevertheless, municipalities have done a lot of work for planning the use of ICT to prepare for future challenges, which can be utilised in the future.

Currently, municipalities are still keen to do most of their ICT-related activities by themselves and avoid joint efforts. Recent research by the Association of Finnish Local and Regional Authorities (Hyvärinen, 2018) clarifies that only 7% of the municipalities made inter-municipal ICT cooperation. At the same time, 35% of the municipalities arranged their municipal ICT by themselves, and 21% bought some parts of their ICT, mainly PC infrastructure, from external vendors and arranged the rest by themselves. In addition to these options, municipalities have established independent ICT companies for their ICT activities, or they have wholly outsourced their ICT.

Although cooperation has been identified as an effective way to take care of difficult public challenges (McCaffrey et al., 1995; O'Leary & Vij, 2012), few municipalities have started ICT cooperation with other nearby municipalities based on their joint need. Since municipalities are independent in their actions, ICT cooperation often starts voluntarily based on individual interests. However, due to limited resources, cooperation without real and short-term benefits will not be accepted, and it will not be continued or even start. To get the best outcome from the ICT cooperation, municipalities should have a clear strategic intent for cooperation and a mutual understanding of the type of joint ICT or the joint ICT structure, which will be needed.

Globally, research for general inter-municipal cooperation has provided interesting findings. For example, findings on cost savings (e.g. Bel & Warner, 2015), better use of limited resources (e.g. Hophmayer-Tokich & Kliot, 2008) and need for better coordination between municipalities (e.g. Rayle & Zegras, 2013) have been reported. Cooperation regarding various e-Government initiatives has also been studied actively (e.g. Layne & Lee, 2001; Pardo, Gil-Garcia & Burke, 2006; Gil-Garcia, 2012; Flak & Solli-Saether, 2013; Guha & Chakrabarti, 2014). However, research for interorganizational ICT cooperation among municipalities is still quite rare. (Moon, 2002; Päivärinta & Dertz, 2008; Sorrentino & Ferro, 2008; Ferro & Sorrentino, 2010; Juell-Skielse et al., 2017). Päivärinta and Dertz focused on IT benefits management in Norway (Päivärinta & Dertz, 2008) while Moon focused on the e-government at the municipal level in the US. Ferro and Sorrentino discussed inter-municipal cooperation in Italy and identified five different forms of municipal cooperation (Sorrentino & Ferro, 2008; Ferro & Sorrentino, 2010), which divided the cooperation based

on their status and goals. The five different forms of cooperation are Convention, Framework agreement, Consortium, Limited company and New public body. A Convention means an agreement between public organizations, where the participants agree on the nature of cooperation while a Framework agreement may be, e.g. a joint supply contract. A Consortium is a legally defined form of cooperation with strict regulations for its members. A Limited company and a New public body are independent organizations used by public organizations to fulfil their separately defined tasks. For the Swedish context, the Italian cooperation forms were used (Juell-Skielse et al., 2017), but slight differences were identified. Based on the Swedish research, they proposed a concept of a cooperation model, which includes the forms of cooperation with assumed benefits while participating in a form. In the Swedish case, the Convention form was using a standard specification instead of loose cooperation. The Swedish Consortium mode combines parts of the New public body, and the Consortium (Ferro & Sorrentino, 2010) added with some software service.

The current thesis aims to clarify, how inter-municipal issues such as the reasons for ICT cooperation, the best type of structure for ICT cooperation and the need for a joint, official IT governance have turned out in the Finnish context.

1.2 Motivation and Research Gap

The current thesis addresses the ICT challenges for public organisations, particularly municipalities, i.e. local governments. Municipalities, especially small and medium-sized municipalities, run their ICT services with very limited resources. A small municipality may have no full-time ICT experts in its payroll, and even a medium-sized municipality may only have two to five full-time ICT experts (Dahlberg & Helin, 2017). Nevertheless, several municipalities try to cope with their ICT challenges alone, without any support or help from other municipalities. It seems that several municipalities have not realised the value of ICT for their daily activities. Without cooperation and ICT resource sharing, municipalities may lose the power of joint purchases and the benefits of optimised operations, centralised problem solving, and even possibilities to hire new competent ICT specialists. Digitalisation and e-service development can also be topics in which ICT cooperation with nearby municipalities could help. How much would well-structured inter-municipal ICT cooperation and joint IT governance help to achieve better IT performance for municipalities? This critical question serves as a motivator for our research.

The current thesis also examines the topicality of the relevant theories and the existing research for IT governance. The amount of IT governance research has

grown during the last decade, reflecting the growing interest in and importance of IT governance in recent years. Simultaneously, the operating environment to enterprises and public organisations has changed dramatically. In the past, organisations typically owned their ICT systems and services and operated the ICT themselves, taking care of problems as they arose or addressing the need for development; however, today, this is no longer possible. Technological challenges, new applications, and internal and external pressures for development are often too broad to be handled by a single organisation. This is why organisations are increasingly creating alliances and organisational networks, or they conduct other types of cooperation activities to stay competitive or even survive. However, IT governance research is still mostly done in the intra-organisational environment focusing on the organisation's internal issues (Brown & Magill, 1994; Brown, 1997; Van Grembergen et al., 2004; Brown & Grant, 2005; Xue et al., 2008). While the trend among organisations is favouring inter-organisational cooperation, more research for inter-organisational ICT cooperation is needed. The current state of IT governance research reveals a research gap concerning whether the current IT governance theories which focus mostly on intraorganisational IT governance issues could be extended to cover inter-organisational IT governance issues. This issue raises another motivator for this thesis.

As explained in Chapter 1.1, another motivating factor for this thesis has been the planned social welfare and healthcare reform in Finland. At the time of writing this thesis, the promotion of the reform was still pending. However, owing to the planned reform, broad interest and pressure towards municipal ICT cooperation and joint IT governance have become critical issues in helping municipalities cope with their future challenges. Even without the reform, the interest in the development of intermunicipal ICT has gained new support, thus providing another motivator for our research.

1.3 Research Objectives and RQs

This thesis aims to clarify two objectives. The first objective of the thesis focuses on practical issues of municipal ICT cooperation. As municipalities are independent in their decision-making, issues concerning inter-municipal cooperation create challenges. This thesis thus examines the reasons why municipalities choose to start ICT cooperation and the benefits they expect to receive. The forms of ICT cooperation and IT governance vary in different patterns of municipal ICT cooperation. Because this can have effects on the results of collaboration, the thesis also examines topics such as joint IT governance and the structure of ICT cooperation.

The second objective of the thesis focuses on theories used in intra-organisation (i.e. within an organisation) ICT research and their usefulness in the inter-organisational (i.e. between organisations) context. Existing theories such as the resourcebased view (RBV) theory (Barney 1991) and the transaction cost economics (TCE) theory (Coase, 1937; Williamson 1975) are often used in the IT governance research focusing on cooperation within one organisation. In this thesis, we aim to clarify whether these theories can be expanded to an inter-organisational cooperation context. We include TCE theory because it focuses on cost reduction, and we use RBV because it focuses on value increases as the reasons and outcomes of governance. We also include the IT governance practices to determine which inter-organisational governance practices are useful in seeking inter-organisational ICT cooperation benefits. We also aim to clarify the relationship between IT governance practices and the benefits of IT cooperation to understand the interaction between these factors. For this, we studied Granovetter's (1973) social network theory to understand what type of meaning the ties (connections) between social groups have for the IT governance practice deployment and the benefits available from the ICT cooperation.

Based on the abovementioned issues, the objective of this thesis is to increase our knowledge about the inter-organisational ICT cooperation between Finnish municipalities. Also, the aim is to understand the reasons for the municipal collaboration and identify the ICT cooperation benefits and structures and the procedures used to achieve those benefits. In addition, we aim to clarify whether the current theoretical basis for the intra-organizational ICT cooperation can be used in the inter-organisational ICT cooperation context.

To fulfil the research objective, we ask three RQs. The first and the second RQs focus on the practical part of the research objective and the third RQ aims to clarify the theoretical part of the research objective.

In the country of the research (Finland), there are over 300 independent municipalities, which have the legal right to choose their methods of operations, which includes deciding whether to cooperate with their nearby municipalities on ICT issues. As the municipalities vary in size, the number of inhabitants, and their industrial structures and locations, the reasons for cooperation and the number of investments for inter-municipal ICT cooperation vary. RQ1 and RQ2 aim to establish the goals and practices for inter-municipal ICT cooperation:

RQ1 What reasons do municipalities have for inter-municipal ICT cooperation? RQ2 How is the inter-municipal ICT cooperation governed?

The TCE, RBV, IT governance practices and Granovetter's (1973) social network theories have been widely used to describe intra-organisational cooperation. To determine whether the abovementioned theories should be used to describe ICT cooperation between organisations, we asked RQ3:

RQ3 What are the possibilities to explain IT activities between organisations using existing theories about an organization's internal IT governance principles and IT governance benefits?

RQ3 will test the contribution of TCE, RBV, IT governance practices and Granovetter's (1973) social network theories in the inter-organisational ICT cooperation environment. RQ3 focuses on inter-organisational cooperation and underlines the importance of ICT cooperation between organisations, i.e. municipalities, in this context.

The thesis focuses on municipal ICT cooperation. However, inter-organisational ICT cooperation in the social welfare and healthcare sector has been included in some of our articles because, in some cooperation studies, healthcare units owned or governed by municipalities were also represented.

1.4 Overview of the Data Collection and Articles

This chapter presents an overview of the data collection and introduces the articles. The relations between RQs and related articles will also be clarified.

1.4.1 Data sources, data collection and data processing

In the thesis, all but one article use case studies to determine how and why IT cooperation has been applied in various municipality groups and how theories can be used to explain inter-organisational IT cooperation.

During the data collection period (2015), there were 317 municipalities in Finland, many of which belonged to different municipality groups. To collect research data through interviews, we searched for municipality groups that were focusing on ICT issues in Finland. The municipality groups which were included in the interviews were located in southwestern Finland (Articles III and V), in Pirkanmaa in southern Finland (Article V) and Uusimaa in southern Finland (Article V). Each municipality group consisted of six to eight municipalities, and the number of residents living in these municipalities was between 100k and 360k. The author of the thesis

organised and conducted the interviews (Kvale, 2006) during 2015. The interviewees were officials who were responsible for the ICT-related issues in the municipality or organisation. An interview was also conducted to complement secondary data for the case study of northern Finland (Article III). Table 1 presents information on the dates of the interviews and the interviewees.

Table 1 Interviews and Interviewees

Interviews	Interviewees	Municipality group
27.2.2015	IT service director	south-western Finland
10.3.2015	IT manager	south-western Finland
11.3.2015	IT manager	south-western Finland
12.3.2015	IM director	south-western Finland
13.3.2015	IM manager	south-western Finland
16.3.2015	IM manager	south-western Finland
16.3.2015	IM manager	south-western Finland
24.3.2015	IM director	south-western Finland
17.6.2015	IM director	Uusimaa
15.12.2015	IM director	Pirkanmaa
23.12.2015	ICT director	northern Finland

The same semi-structured questionnaire was used in all the interviews with a list of prepared questions which guided the discussion. Subjects outside of the list were also discussed if interesting issues came out. The interviews, which lasted an average of one and a half hours, were recorded and transcribed according to a transcription reference style. The questionnaire used in the interviews is presented in the Appendix.

The interview questions were targeted towards four core interest areas:

- a) The background of the ICT cooperation to improve the understanding of why and how ICT cooperation has started
- b) The practice of the ICT cooperation to collect information about how ICT cooperation is taking place.
- c) The impact of the ICT cooperation to clarify the effects and possible benefits and challenges of ICT cooperation.
- d) The future of the ICT cooperation to identify the visions and development ideas of ICT cooperation.

Data gathered in Northern Ostrobothnia was based on a study conducted by theother researcher and co-author for Articles II, III and IV. The study follows the formation of IT governance for social welfare and healthcare organisations covering over 100 different organisations including 68 municipalities, five social welfare development districts, five healthcare districts, 33 healthcare centres and nine hospitals in the

Northern Ostrobothnia area. The data were complemented with a nationwide, webenabled survey emailed to 260 social welfare and healthcare experts to evaluate the proposed arrangement.

The self-collected data were complemented with secondary data from other sources. This secondary data is also based on interviews conducted by other researchers or consultants. The detailed structure or the exact content of these interviews was not available, but the outcomes and the results of the interviews were at our disposal for use as secondary data. The municipality groups which were included in the research using secondary data are located in northern Finland (Article III), Central Finland (Article IV) and nationwide (Articles III, and IV). In the study of northern Finland, nine municipalities and one joint municipal authority in the northern part of Finland were included, and the Central Finland study included 23 municipalities in Central Finland. In the nation-wide study, 144 municipalities from 20 municipal regions were listed, but in the end, only 17 municipal regions were included due to different report contents. These reports were focusing on details not crucial for our study.

In addition to the field data collection, a literature review was conducted to clarify the current state of IT governance research. The literature review is published in Article I. The content and the information of Article I is used as supporting material, adding value to all articles replying to the three RQs.

This thesis aimed to clarify the reasons for inter-municipal ICT cooperation and implementation for inter-organisational IT governance. Articles II, III, IV and V present the research conducted for this topic. Another aim of this thesis was to clarify the form and function of inter-municipal ICT cooperation, and Articles II, III and V present the research for this topic. This thesis also clarifies if the theories and IT governance principles used in the intra-organisational context, i.e. IT governance inside an organisation could also be used in the inter-organisational context. Articles I, II and IV discuss this issue.

Table 2 presents the relations between the Research Questions (RQs) and the articles. The articles which provide relevant answers to the RQs are marked with an 'X'.

 Table 2
 Research Questions and Related Articles

Article	RQ1	RQ2	RQ3	
Article I			X	
Article II	X	Χ	X	
Article III	X	X		
Article IV	X		X	
Article V	X	X		

1.4.2 The articles included in the thesis

The five articles included in the thesis are as follows:

Article I 'Inter-Organizational IT Governance Research: A Literature Review' by Ari Helin (2019).

In this article, we investigate what has been studied about the IT governance between different organisations. This article gives general backing to research for inter-organisational IT governance and provides answers to RQ3.

- Article II 'Formation of Voluntary Inter-Organizational IT Governance for Healthcare and Social Welfare IT Theoretical Background and Empirical Evaluation' by Tomi Dahlberg and Ari Helin (2014). This article investigates the formation of IT governance among municipalities, healthcare and social welfare organisations. We examine the reasons to start a cooperation and the accepted IT governance arrangement. We also examine whether the theories used in the intra-organisational context can be used to explain cooperation in the inter-organisational context. This article finds answers to all three RQs.
- Article III 'Volume, Benefits and Factors that Influence Inter-municipal ICT Cooperation in Relation to ICT-related Social Services and Healthcare Services' by Ari Helin and Tomi Dahlberg (2017). In this article, we evaluate how ICT cooperation is carried out in several Finnish municipal regions. For the research, we used nationwide research and three regional studies. The article finds answers to RQ1 and RQ2. This article is an extension of a conference paper (Dahlberg & Helin, 2016).
- Article IV 'How IT Governance Practices Contribute to Inter-Municipal ICT Cooperation and Its benefits: Indeed, "The Emperor Has No Clothes" by Tomi Dahlberg and Ari Helin (2017).

 In this article, we continue the research of several Finnish municipal regions and link the theory-proposed cooperation benefits to practices detected in the municipal regions. We also extend the theory base of IT governance research to inter-organisational context and show how this theory base can be used empirically. The article also

confirms the role and importance of expectations and benefits which public organisations are looking to establish through ICT cooperation. The article answers RQ1 and RQ3 and is an extension of a conference paper (Dahlberg & Helin, 2017).

Article V 'SECOs as a Mean to Survive? – Case Municipal ICT' by Ari Helin (2017).

In this article, we use empirical research to examine different ICT operating models for inter-municipal ICT cooperation and its impact on the success of ICT cooperation in three different regional settings. The article provides answers to RQ1 and RQ2.

The articles included in the thesis focus on the topic of municipal ICT cooperation and IT governance. Additional articles discussing the same topic have been created by the author and his colleagues, which include some common elements, e.g. arguments and discussions. The full list of articles is presented in the thesis under the List of Publications.

1.5 The Structure of the Thesis

The thesis consists of two parts. Part One is divided into five chapters. Chapter 1 presents the background to the research, introduces the motivation and the research gap, presents the research objectives and the RQs, gives an overview of the data collection and the articles included in the thesis, and introduces the structure of the thesis. Chapter 2 presents the theories used in the thesis, and Chapter 3 introduces the methodology and the process used in the data collection. The author's role in the thesis is also clarified. Chapter 4 presents the articles used in the thesis and discloses the research results. Chapter 5 explains the theoretical and practical implications of the thesis, discusses the limitations of the current research and provides suggestions for future research. Part One concludes with a summary of the thesis.

Part Two contains the research articles and appendix. There are five research articles, including three journal articles (Articles I, III and IV) and two conference papers (Articles II and V).

2 Theoretical Background

This chapter reviews the theoretical background and clarifies the theories, frameworks and definitions used in this thesis. Two theoretical entities (IT governance and inter-organisational cooperation) are considered as the main theories in the thesis. The theories have been chosen due to their extensive use in IS research and due to their expected and tested suitability to network behavioural research as supported by Walsham (Walsham, 2006).

2.1 IT Governance

The term 'IT governance' first appeared in academic discussions during the 1990s (Loh & Venkatraman, 1992; Henderson & Venkatraman, 1999). IT governance is defined as 'specifying the decision rights and accountability framework to encourage desirable behaviour in the use of IT' (Weill and Woodham, 2002, p. 1). This was later amended to 'IT governance is the process by which firms align IT actions with their performance goals and assign accountability for those actions, and their outcomes' (Weill and Ross, 2004, p. 1). IT governance is defined by the ISO/IEC standard as a 'system by which the current and future use of IT is directed and controlled' (ISO/IEC 38500, 2015, p. 3).

To identify the current state of IT governance research, we examined the relevant literature. Many definitions, frameworks and general studies have been made in the last two decades. Dahlberg and Kivijärvi (2006) presented an IT governance framework. They introduced an assessment tool to measure its effectiveness, and Peterson (2006) presented a holistic view of IT governance using an IT Governance Assessment Process (ITGAP) model which can be used to measure the effectiveness of IT governance architecture. Van Grembergen and De Haes (2012) focused on essential theories and practices around IT governance, which created the base for their research and practical experiences. Van Grembergen et al. (2004) claimed that the execution of governance takes place through the structure, process and relational mechanism practices. In addition, research related to existing IT governance practices and their impact on performance impact (Ali & Green, 2009, 2012), the maturity of busi-

ness-IT alignment (De Haes & Van Grembergen, 2009) and organisational performance (Turel & Bart, 2014) has been conducted. Research related to the most widely used IT governance practices (De Haes & Van Grembergen, 2009; Almeida et al., 2013) provides a good list of commonly used IT governance practices. The formation of IT governance has been presented in various best practice collections and frameworks, including ITIL (2018), TOGAF 9.2 and COBIT 5. ITIL focuses on IT service management practices while TOGAF is a framework for enterprise architecture. COBIT is a framework for the governance and management of enterprise IT. According to the abovementioned standards, IT governance is part of enterprise governance. In practice, this means that ICT cooperation should bring value to the participating organisations. The best practice methods describe the objects of ICT cooperation which, for municipalities, includes the ICT services they should incorporate into their ICT cooperation. Existing literature (Das & Teng, 2000; Melville, 2004; Park et al., 2004; Wiengarten et al., 2013) have presented some valuable constructs for understanding and delineating inter-organisational IT governance principles and benefits.

Research for IT governance has been rather widespread. IT governance research has generally focused on ICT issues which usually take place inside an organisation. Some studies have focused on IT governance in general (e.g. Anica-Popa, 2012; Leclercq-Vandelannoitte, 2018; Héroux & Fortin, 2013, 2018; Turel & Bart, 2014) while some studies have focused on IT governance framework models such as ITIL (e.g. Iden & Eikebrokk, 2014; Nicho & Al Mourad, 2012) or COBIT (e.g. Abu-Musa, 2009; De Haes & Van Grembergen, 2013; Souza Neto, 2013). A large number of IT governance studies have focused on enterprise IT governance. Some are considered from the audit, or assessment perspective (e.g. Bin-Abbas & Bakry, 2014; Gheorghe, 2011, 2010; Iliescu, 2010) or they take the IT governance improvement perspective (e.g. De Haes & Van Grembergen, 2013; Lombardi et al., 2016; Lorences & Avila, 2013).

There are also IT governance studies, which focus on public organisation IT governance. Although challenges for managerial-level IT issues are similar in the public and private sectors, there are systemic differences between organisations (Campbell et al., 2009) which must be taken into account. Studies for IT governance for public organisations often focus on hospitals (e.g. Bradley et al., 2012; Köbler et al., 2010; Lagsten & Nordström, 2015; Gebre-Mariam & Fruijtier, 2018; Smith et al., 2013) or more widely on public sector organisations (e.g. Ali & Green, 2007; Heindrickson & Santos, 2014; Nfuka & Rusu, 2013; Dawson et al., 2016; Pang, 2014).

Relatively few studies deal with inter-organisational IT governance. For example, Croteau and Bergeron (2009) defined various modes of IT governance that contribute to successful inter-organisational relationships. Prasad et al. (2012) examined

different collaborative organisational structures and identified suitable IT governance structures for them. Zarvic et al. (2012) presented practices for inter-organisational IT governance, and Croteau et al. (2013) identified critical inter-organisational IT governance mechanisms in use. Xiao et al. (2013) investigated the buyer-supplier dyadic relationships and the impact of inter-firm IT governance on information sharing, and Chi et al. (2017) studied the relationship between two inter-firm governance strategies (balancing and complementing governance strategies) and their impact on relational performance.

2.2 Inter-organisational Cooperation Framework

Existing ICT cooperation research in information systems (IS) focuses mainly on intra-organisational cooperation, i.e. cooperation within an organisation. This thesis extends the existing research theme for collaboration between organisations, i.e. municipalities. To identify the current state of inter-organisational ICT cooperation, we examined the relevant literature in Article I.

Inter-organisational cooperation has been widely studied in various business areas such as economics, strategic management, organisational behaviour and sociology of organisations (Borgatti et al., 2003; Zaheer & Venkatraman, 1995; Zaheer et al., 2010). However, in IS research, research focusing on inter-organisational research has been limited. The primary research interest has been towards IT cooperation within one organisation (e.g. Van Grembergen et al., 2004; Van Grembergen & De Haes, 2008; Weill & Ross, 2004). Nevertheless, reasons for cooperation in an inter-organisational structure are often the same as those in an intra-organisational structure, mainly targeting better efficiency and optimised costs.

In Nordic countries, little research has focused on municipal ICT cooperation. In Finland, few research studies could find systematically organised ICT cooperation between municipalities (Hyvönen, 2015; Dahlberg, 2016). In Sweden, a study for cooperation among public organisations around digital archiving systems (Juell-Skielse et al., 2017) discovered five different modes of cooperation with various benefit expectations. A study in the municipal sector in Norway (Päivärinta & Dertz, 2008) identified a set of pre-determinants affecting IT benefits management practices. Overall, in the context of the thesis, inter-organisational IT governance is seen as an arrangement to govern IT as a joint activity.

Several theories and frameworks explain cooperation between organisations and the benefits of collaboration, including a knowledge-based theory of inter-firm collaboration (Grant & Baden-Fuller, 1995) and a dynamic capabilities framework (Teece et al., 1997). However, according to Ebers (1997), organisational cooperation

usually provides two types of benefits: revenue increases and cost reductions. The theories which are typically followed are the resource-based view (RBV) theory (Barney, 1991; Wiengarten et al., 2013) and the transaction cost economics (TCE) theory (Coase, 1937; Williamson, 1975, 1985; Geyskens et al., 2006). The RBV theory addresses adding value and revenue increases from configurations of resources and TCE theory addresses cost reduction from transactions, which are both crucial issues in the municipal sector due to the tightening economic situation. These theories have been actively used in IS research (e.g. Park et al., 2004; Magni et al., 2012-2013; Wiengarten et al., 2013). RBV and TCE separate governance into three levels: single organisation, market and alliance/network (Williamson, 1985; Gulati, 1999; Das & Teng, 2000; Zaheer et al., 2010) and they focus on three themes: alliances, cooperation and their governance. These theories also clarify the link between interorganisational IT governance and the ICT cooperation benefits.

Inter-organisational ICT cooperation and joint IT governance take place between people in social groups. Thus, it should be clarified how the relations between social groups impact ICT cooperation and IT governance practice usage. Granovetter's (1973) social network theory offers a theoretical basis for explaining the impact of social contacts (i.e. ties) on cooperation. The social network theory is often used because it describes the connections between organisations in networks (Borgatti & Foster, 2003). RBV, TCE and Granovetter's (1973) social network theory is therefore selected for this thesis and are explained in more detail in the following sections.

2.2.1 RBV theory

The RBV approach surfaced in 1959 by E. T. Penrose. Since then, many alternative and complementary models have been developed based on the RBV approach and the research tradition. Barney (1991) introduced RBV to information systems research, which has extensively been used in IS research with a focus on IT business value research (Wiengarten, 2013). RBV keeps every organisation unique in terms of its value creation potential (Barney, 1991; Wiengarten, 2013). Barney (1991) created the VRIN model defining four resource properties which make the resources unique: Valuable, Rare, Imperfectly imitable and Non-substitutable (Barney, 1991). For example, in the municipality context, these may be interpreted as follows:

- Valuable: a municipality can have more tax-paying companies than other nearby municipalities;
- Rare: a municipality has the only university in the area;
- Imperfectly imitable: a municipality has a longer coastline than other nearby municipalities; and

- Non-substitutable: a municipality has the only harbour in the area. Many value-adding indicators have been identified by RBV research (Wiengarten, 2013; Das & Teng, 2000; Park et al., 2004).

An organisation (a municipality in the context of the thesis) can be described as a wide range of tangible or intangible resources or assets, owned by the organisation or to which the organisation has access at least semi-permanently (Das & Teng, 2000; Geyskens, 2006; Wiengarten, 2013). The value creation potential of an organisation is defined by the organisation's resources or assets, especially those which are imperfectly mobile, non-imitable and non-substitutable (Barney, 1991). The arrangements of how to use resources are included. The following alternatives for the resource usage are considered: own production, sourcing from markets and cooperation. Measures of the value increase identified by RBV research include conserving resources, sharing risks, obtaining information, accessing complementary resources, reducing (product/service) development costs, improving technological capabilities and enhancing reliability (Park et al., 2004).

According to RBV, inter-organisational cooperation is worth implementing if the participants receive more value through cooperation and if the value cannot easily be created through any other governance or operating model. This cooperation may take place through pooling, aggregating, sharing or exchanging their unique resources. According to RBV, the willingness for cooperation is also affected by the trust or lack of trust (Osborn, 1990; Zaheer & Venkatraman, 1995; Das & Teng, 2000; Gereffi, 2005; Park et al., 2004). Trust is an essential requirement in successful inter-organisational IT governance.

RBV focuses on how to increase the value of transactions by using the available resources (Wiengarten, 2013). For example, in the context of the thesis, how can the value and availability of municipal ICT services be increased? According to RBV, there are three different governance models: single organisations, markets and alliances (or networks). The target for alliances, which are voluntary cooperative interorganisational agreements, is to obtain a competitive advantage for the participants of the alliance (Das & Teng, 2000). IT resources may deliver value to an alliance by creating synergies and higher-level capabilities, enabling long-term performance improvement and demanding a re-design for human resources and facilities (Wiengarten, 2013). IT resources, which can be divided into human and technical resources (Das & Teng, 2000; Melville, 2004), are usually complementary to an organisation's other non-IT and IT resources (Wiengarten, 2013).

In the municipal context, RBV helps to consider options for IT activities. Should a municipality manage all its IT operations alone, without cooperation? Would cooperation with nearby municipalities add value and increase service potential? In case the IT cooperation is selected, to what extent is it done and who will be the optimal partner for the cooperation? Should IT activities only be outsourced and

bought from an external vendor? These are difficult questions faced by every IT manager in municipalities.

2.2.2 TCE theory

TCE theory was introduced in 1937 by R. H. Coase, who used governance structure to describe markets and hierarchies. Differences in transaction costs define the choice between markets and hierarchies more than production costs (Geyskens, 2006). Williamson (1975, 1985) developed TCE by stating that when organisations make decisions about whether to make, buy or ally, they use asset specificity, uncertainty and transaction frequency as the determinants.

The basis of TCE is that, in executing its transactions, an organisation should rely, if possible, on a form of governance that involves the least transaction costs (Williamson, 1985; Geyskens et al., 2006). Municipalities can save costs through ICT cooperation by sharing their limited ICT resources or by reducing their ICT service-related risks. The form of governance should be the best offering with the highest cost savings for all ICT cooperation participants. Various cost-saving measures have been recognised in research (Geyskens et al., 2006).

TCE theory also addresses issues such as where the economic transaction should take place and who should be conducting the transactions. In the municipal cooperation context, this refers to which municipal is the best actor to conduct a specific ICT transaction and where the transaction should take place. Geyskens, Steenkamp and Kumar (Geyskens et al., 2006) introduced two forms of networks/alliances into TCE theory: hierarchical governance and relational governance. Hierarchical governance is the structure in which one partner has the authority to develop rules, give instructions and impose decisions on the other partners. This mode of governance is typical for supply and value chains, where a company controls other partners (Gereffi et al., 2005). Relational governance is a joint activity in which the parties involved jointly develop policies to achieve the defined goals (Geyskens et al., 2006). Transactions between a leading organisation and other partners create hierarchical networks while transactions in networks and alliances form relational governance or networked governance (Williamson, 1985; Wiengarten, 2013). Municipal inter-organisational ICT arrangements, which have been studied throughout the thesis, are relational hierarchies. In relational governance, the role of trust is vital (Geyskens et al., 2006). Trust is needed in the establishment and during the execution of ICT cooperation. Lack of trust has a detrimental effect on ICT cooperation and lowers the results.

According to TCE theory, economic transactions inside an organisation are considered hierarchical governance while transactions between two parties are called market governance (Coase, 1937; Williamson, 1975). Market governance works best when assets used in executing transactions are similar, when the use of assets is large or very small and/or when asset usage uncertainty can be reduced (Williamson, 1985; Geyskens et al., 2006). In the context of a municipal ICT environment, this means that best results will be reached when all participating municipalities have roughly similar ICT services, when the volume of their ICT services is high (or seldom used and expensive) and/or when their ICT cost uncertainties can be reduced by sharing.

TCE provides tools for cost optimising for municipalities. Municipalities should consider which activities are better to keep for themselves and which tasks are better to share or outsource. Municipalities should also determine which of the nearby municipalities would be the best partners for cooperation regarding resources, IT volumes and the type of services provided. While such questions may be difficult, the right conclusions will provide better results for their IT cooperation.

2.2.3 Social network theory

Granovetter (1973) first introduced the social network theory in the 1970s to examine the labour market and study the social processes of job-hunting. The theory was later expanded to macro-level structures in studies for the social embeddedness of institutions (Granovetter, 1985). At that time, it was also noted that markets consist of social rather than economic structures (Baker, 1984). Since that time, social network research has been used in many research areas and disciplines. The social network theory is sometimes seen more as a research approach than a theory (Bögenhold, 2013).

The primary construct in the social network theory is the tie construct (Granovetter, 1973, 1985). The ties describe the information flow between individuals in the same or different social groups, and they enable the trust to be built between groups and individuals (Granovetter, 2005). The ties can be divided into strong tie constructs and weak tie constructs. With the strong tie construct, the social relations inside a social group are meaningful. In a municipality, officials discuss various issues regularly with each other and share the same values, norms, objectives and other similar factors. While doing so, a municipality's identity strengthens, thus creating strong ties within the municipality. In the weak tie construct, the social relations between different social groups are meaningful. Individuals belong to various social groups, where they meet individuals outside their regular social group and discuss and share ideas. This social contact is important because the flow of ideas takes place more

through weak ties than through strong ties (Granovetter, 1985, 2005). Without contacts between individuals from different municipalities (i.e. weak ties), ICT cooperation between municipalities may be scarce or missing entirely. Respectively, if contacts between different municipality groups are frequent and extensive, ICT cooperation may prosper and bring benefits.

The network density and the structural hole (Granovetter, 1973, 1985, 2005) are other constructs related to the weak and strong ties. The strong ties are tight when the network's density is high. However, the density is often low in large networks. A structural hole may exist if there are too strong ties in the network or if the weak ties are missing in the network. This structural hole may be filled by an outsider, who will take care of inter-organisational contacts (Granovetter, 1985, 2005).

Other important constructs in social network theory are the embeddedness of organisations (Granovetter, 2005) and social capital. Embeddedness affects the actions which the organisations take in their operations. The history of relationships between organisations is essential. Depending on the length and scope of the relationships, the actions tend to differ (Zaheer et al., 2010). This social capital defines the value of inter-organisational connections.

Unlike in ICT cooperation, where most of the activities are measured by economic terms, social relationships are mostly non-economic (Granovetter, 1973, 2005). Hence, social indicators should also be included in the measurement of ICT cooperation. The social context may also change the actions taken by organisations (Gulati, 1999).

The social network theory clarifies many important constructs, which helps to understand inter-municipal ICT cooperation. Because of the legal rights for Finnish municipalities to decide and manage their ICT activities and actions independently, differences in operational working ways and wishes for ICT cooperation exist. Small municipalities act as an example here since municipal officials often have strong ties with their local colleagues belonging to the same municipality. Small municipalities may be suspicious about the ideas and actions provided by other municipalities, especially if the nearby municipality is far bigger. Officials in a small municipality may fear that ICT cooperation is the first step towards a merger, which is why all cooperation efforts will be avoided or attenuated. In doing so, they lose all the benefits that a fruitful ICT cooperation may bring.

2.3 Resource Dependency Theory

In addition to the abovementioned theories, this thesis includes another theory because municipalities which take independent actions based on the Constitution are often forced to cooperate to succeed in their activities. The theory, which explains the reasons for this cooperation, is the resource dependency theory (RDT) (Pfeffer & Salancik, 1978). RDT focuses on dependencies between actors. Actors (municipalities, in the context of the thesis) depend on other actors unless they have full control over the activity. RDT states that most of the organisation's results are achieved through a joint operation of interoperable actors. In modern, Western societies, such as countries like Finland, where organisations often cooperate in many different fields to succeed, resource interdependence is a typical way of working. Interdependence should include ICT functions which regularly face resource constraints.

In the municipal context, local ICT managers are facing challenges brought about by efficiency pressures. Often the problems are caused by a large number of external requirements coming from the national government, other administrative authorities or from residents related to new ICT services and applications. Not all of these requirements can be fulfilled due to limited internal resources. These limitations cause challenges for ICT operations and the development of municipal ICT activities. As a solution, some ICT managers have started ICT cooperation with nearby municipalities to extend their limited ICT resources and satisfy the need for new services. With the help of ICT cooperation, ICT managers are able to get support from their colleagues, discuss and share opinions, receive training and consultation, and share resources when possible. Understanding the resource dependencies will thus encourage inter-municipal ICT cooperation.

3 Research Approach and Procedure

This chapter presents the methodology and the research methods used in this thesis. The implementation of the methods for each case is explained separately.

3.1 Methodology

The core of the thesis is to evaluate municipalities in their daily ICT operations with their nearby municipalities. According to the Constitution (Ministry of Justice, 1999), municipalities have the right to decide how they will fulfil their duties towards their residents, taking into account the limitations mentioned in the Constitution. Municipalities differ in size, location and financial possibilities and their residents have different opinions and ideas about how the authorities should manage municipal ICT; thus, various ways to implement municipal ICT operations exist.

The epistemology of the research can be positivist, interpretive or critical (Chua, 1986). In the positivist studies, the research target is assumed fixed and not dependent on people being active and decisive about their affairs (Orlikowski & Baroudi, 1991). In critical studies, a researcher attempts to evaluate and change the research environment critically. In interpretive studies, the world is not fixed, and the researcher attempts to understand how people's actions and decisions affect their environment (Orlikowski & Baroudi, 1991). Also, interpretive research supports IS researchers in understanding human thinking and actions in organizational and social contexts (Klein & Myers, 1999). As Orlikowski and Baroudi (1991) stated, interpretive research in IS supposes the world to be non-standard, where there is always room for interpretation. The goal of interpretivism, then, is to understand the subjective meanings of people (Goldkuhl, 2012). In interpretivism, the theory is used to explain actions. With all actions, there is a meaning and some intention based on social and historical practices (Chua, 1986). The role of the researcher is the outside observer without direct involvement in the focus area (Wahlsham, 1995, 2006). In the research, interviews are actively used as the mean of information collection supplemented by other forms of data when possible. The target is not to report facts but to report the interpretations of the researcher of the interviewees' interpretations. In

the case of municipal ICT operations, where people in charge make decisions about the actions of their municipalities, neither a positivist nor critical approach is feasible. Therefore, the epistemology of this research is interpretivism.

The ontology of interpretive IS system research presumes that the social world is created by people through their actions (Orlikowski & Baroudi, 1991). Organisations and inter-organisational cooperation do not take place unless a living person works to make it possible. People have different ideas, thoughts, comments and decisions; they construct their reality (Walsham, 1995). These differences also affect the expectations for the benefits of municipal ICT cooperation and the implementation of the cooperation. Inter-organisational ICT cooperation between municipalities thus has different embodiments. Therefore, the ontology of the thesis is to understand the municipal ICT operations through real cases and to clarify the reasons for inter-municipal ICT cooperation.

This thesis uses empirical field information collected through various case studies around Finland; these are mainly from municipalities but also healthcare organisations. As the primary research method, interviews were conducted by the author of the thesis or by external consults or other researchers. Also, various documentation, project-meeting memos, participation in the meetings and secondary data have been used for information collection. While research methods have to provide support for finding answers to RQs (Venkatesh et al., 2013), the usage of various research methods is valid. Additionally, according to Mingers (2001), research results may be more valuable and trustworthy when different research methods are used. As Mason (2006) argued, mixing methods provides excellent possibilities for us to understand the social experience better and improve our capabilities for the explanation and generalisation. Thus, for this study, the selection of multiple methods as the research approach is appropriate (Venkatesh et al., 2013).

The case research strategy (Benbasat et al., 1987; Yin, 2009) has been used as the primary research method. According to Yin (2009), the case study is often used in situations where information about groups or organisations is collected. The case study research is usable in IS research for the following three reasons (Benbasat et al., 1987):

- a) IS can be studied in a natural setting to learn about the latest transactions in the field and to create theories from practice;
- b) 'how' and 'why' questions can be asked to understand the complexity and nature of the actions; and
- c) for areas where only a few studies exist, a case research is suitable. These reasons are appropriate for our research context.

As stated earlier, various independent case studies are used to collect empirical information. The findings from the cases have been reported independently. In our research setting, cases around Finland are used to collect data and answer our RQs.

The most used qualitative research method is an interview (Polkinghorne, 2005; Myers & Newman, 2007) and, in this thesis, interviews are used as the primary information gathering method. Interviews make it possible for a researcher to get to know the participants' specific experiences and their feelings about them (Schultze & Avital, 2011). The interviews, which were conducted by the researchers, used an openended, semi-structured questionnaire (Myers & Newman, 2007). The questions were used as a guide and as a checklist for topics to be covered during the interview. Variations were made, and follow-up questions asked depending on the situation and the participants' answers and comments. The questionnaire was pre-tested among qualified researchers and used similarly in all interviews. The names of the interviewees were anonymised to maintain confidentiality. All research data are stored according to the university's guidelines. Data from interviews conducted by external consultants were also collected in the form of research results for use as secondary data.

Case studies can collect data from various sources such as interviews, documentation and observations (Benbasat et al., 1987; Eisenhardt, 1989). In addition to interviews, this thesis uses various other information sources such as project documentation, workshop material, observations and surveys. The information sources are explained in Chapter 3.2 case by case. In Article II, a survey (Pinsonneault & Kraemer, 1993) was also used because it offers the researcher a possibility to complement qualitative research to confirm and quantify the results (Newsted et al., 1998). In Chapter 3.2, where the empirical studies are reviewed, the usage of various data sources is clarified.

3.2 Overview to Research Methods and Research Processes

This section introduces the research methods used in the articles. The research methods for each article are explained separately. If the same case has been used in different articles, its research method has been explained only once.

Article I, 'Inter-Organizational IT Governance Research – A Literature Review', used a systematic, structured literature review process (Webster & Watson, 2002) for data collection. The target of the literature review was to identify the current state of the inter-organisational IT governance among the IS domain. To obtain the latest academic content, the review was limited to articles and conference proceedings published in the last two decades. For the search process, Primo Central Index (Volter, 2018) with references from over 900 databases was used with keywords 'ITG' and 'inter-organizational' and their combinations and synonyms. Additional

search criteria 'Full text available', 'Peer-reviewed', 'English language' and 'The year of publication from 2000' were used.

Further, the top eight journals from the AIS Senior Scholars' Basket of Journals were reviewed. Initially, 210 academic articles and conference proceedings were retrieved. Duplicates, articles with non-matching titles, articles that did not focus on IT governance and articles that did not focus on inter-organisational issues were removed, leaving 26 relevant articles. The article collection was followed by their categorisation and analysis. Finally, the findings were reported.

Article II, 'Formation of Voluntary Inter-Organizational IT Governance for Healthcare and Social Welfare IT – Theoretical Background and Empirical Evaluation', had multiple goals. The aim was to confirm the usability of the RBV theory, TCE theory and Granovetter's (1973) social network theory in an inter-organisational context and to describe the benefits that are available when using the IT governance principles. The study examines the importance of IT governance principles in the formation of IT governance between healthcare and social welfare organisations. The research started with a theoretical background. While understanding that literature on inter-organisational IT governance is rare, we decided to use the principles of the intra-organisational IT governance and extend these into an inter-organisational context with the constructs from selected other theories. Theories related to IT governance, RBV, TCE and social networks were included. RBV was selected because it focuses on revenue/value increases (Wiengarten et al., 2013). TCE was selected because it focuses on cost reduction (Geyskens et al., 2006) as the reasons and outcomes of governance. Social network theory was selected because it depicts the connections between organisations in networks (Borgatti & Foster, 2003).

In the empirical research, the primary research method was the reflective observation approach method (e.g. Kolb, 1984) in a single case context because one of the researchers was acting as a consultant in the case. The case study guidelines (Yin, 2009) and the constructs for building a research study were followed (Eisenhardt, 1989; Sireci, 1998). The full project material, including meeting memos and presentations, was also accessible and provided a comprehensive description of the implementation of the project. In the project, more than 100 organisations were included in the healthcare and social welfare areas in one of the country's five Special Catchment areas covering almost 50% of the country. Observations and documents were analysed and documented. The reflective observation approach was supplemented with a web-enabled self-administered survey (Pinsonneault & Kraemer, 1993). The survey was organised to allow subject experts to evaluate the expected benefits, IT governance principles and the roles of organisations involved. The design of the survey and the survey details were based on general guidance (Eisenhardt, 1989; Sireci, 1998). Only a part of the data which focused on IT governance principles and benefits was used in the current study. In the survey, pre-formulated statements were included, and participants were expected to evaluate statements on a seven-point Likert scale (Allen & Seaman, 2007). Of the 260 invitations that were sent out, 76 responses were received, giving a response rate of 26%. Responses were collected, analysed and documented, and the results from the project work and the survey were collected.

In Article III, 'Volume, Benefits and Factors that Influence Inter-municipal ICT Cooperation in Relation to ICT-related Social Services and Healthcare Services', the purpose of the research was to obtain a broader understanding of the cooperation and to determine how municipalities conduct inter-municipal ICT cooperation. The research process started with a theoretical background. TCE theory (Williamson, 1975) and RBV theory (Wiengarten et al., 2013) were used to describe the benefits available from ICT cooperation. RDT (Pfeffer & Salancik, 1978) and Granovetter's (1973, 1983, 1985, 2005) social network theory were included to clarify the resource, and the social network-related issues affecting ICT cooperation.

For the empirical part, data from a nationwide study and three regional studies were used. The data and results from the nationwide study (Ministry of Finance, 2015) were used as secondary data. The nationwide study consisted of data obtained about ICT usage in 144 municipalities around the country grouped in 20 regional areas. The nationwide study was done by various independent consultants, who were using interviews and other available data to collect understanding about the ICT situation in the respective areas. Based on their understanding, they wrote area-specific reports, which were collected by the Ministry of Finance, which made them public. Owing to the inconsistency of the reports, we were able to use only 17 of the regional area reports. The researchers read and analysed the reports separately, compared the findings and identified the common conclusions.

Data from three regional studies, which consisted of a municipal group from south-western Finland, a municipality and a healthcare organisation group from northern Finland and a municipal group from northern Finland, were also used. In the south-western Finland study, ICT leaders of the group were interviewed by the author of the thesis using a thematic, open-ended, semi-structured questionnaire (Myers & Newman, 2007). Based on the interviews, the researcher wrote a report about his findings.

In the northern Finland studies, data from two case studies (Hyvönen, 2015; Dahlberg, 2016) were used. In the Hyvönen case, it was investigated how ICT services were arranged in the 23 municipalities of one Finnish region by interviewing those responsible for ICT in each municipality. Open-ended questions were used to collect the understanding, how the ICT issues in the municipalities were. Supplemented by relevant municipal data, findings were collected and documented. In the Dahlberg case, observations, documents supplemented by a survey were used as described in Article II. Finally, we collected and documented the findings from all cases.

Article IV, 'How IT Governance Practices Contribute to Inter-Municipal ICT Cooperation and its benefits: Indeed, "The Emperor Has No Clothes", examined how ICT cooperation between municipalities is carried out and governed in 20 municipal regions in Finland. The research process started with the information collection for the theoretical background. TCE (Coase, 1937; Williamson, 1975, 1985) and RBV (Barney, 1991; Wiengarten et al., 2013) were included as explainers for the ICT cooperation, common IT governance as means to cooperation benefits and social network theory as a predictor of cooperation. For the empirical part, the results from existing studies conducted in central Finland (Hyvönen, 2015), northern Ostrobothnia (Dahlberg, 2016) and the nationwide study (Ministry of Finance, 2015) were used. Reports from the nationwide study were read and analysed separately by the researchers; the findings were compared, and the mutual conclusions identified. A comparison was made between the theoretical benefits and governance practices with those we identified empirically from the municipal regions and prior studies. To conclude the research and to obtain a better understanding of empirical findings, we reviewed Granovetter's (1973) social network theory.

Article V, 'SECOs as a Mean to Survive? - Case Municipal ICT', examined whether the municipal ICT cooperation can be considered a software ecosystem (SECO)? The research process started by reading theories about software ecosystems (Moore, 1993; Manikas & Hansen, 2013). In a SECO, actors constituting a proper ecosystem correspond to the participants, which can be identified in a municipal ICT cooperation. In addition, RBV (Barney, 1991) was viewed for cost and resource optimisation reasons, TCE (Williamson, 1975) for cooperation costs reasons and Granovetter's (1973) social network theory for the importance of connections and trust between groups and people. In the empirical part of the research, we collected information from three municipal groups with joint ICT activities. All municipality groups were located in the southern part of Finland covering six to eight municipalities each. The author of the thesis conducted 10 thematic semi-structured interviews among ICT leaders of the municipal groups to get an understanding about the reasons for the ICT cooperation, about the benefits municipalities have received through the cooperation and about the future visions about the inter-municipal cooperation. In the interviews, the same questionnaire (Appendix) was used. The interviews were transcribed, analysed and documented, and the findings of the ICT issues were documented.

3.3 Author's Role in the Articles

This section clarifies the role of the author of the thesis in each article. The role varies from taking a co-author role to that of the sole author.

Article I was produced solely by the author of this thesis, and for Article II, the author focused on the part of the theoretical background. In Article III, the author of this thesis focused on creating a part of the theoretical background, conducting interviews in the south-western Finland study, collecting and analysing the data for the nationwide study and collecting data from northern Finland's municipal study. The author of this thesis also performed practical work related to the publishing process. In Article IV, the author created a part of the theoretical background, and performed the data collection and analyses of the nationwide study. The author also completed a part of the writing. Article V was produced solely by the author of this thesis.

4 Empirical Studies and Results

This chapter introduces the articles included in the thesis with the findings. The chapter also presents the results and reveals the connections between the articles and the ROs.

4.1 The Articles

The following sections present all five articles included in the thesis. Each article is presented with the research findings. The role of the article for this thesis is also clarified.

4.1.1 Article I

The article 'Inter-Organizational IT Governance Research: A Literature Review' by Helin (2019) clarified the current state of the inter-organisational IT governance research.

The findings of the literature review highlight that the current inter-organisational IT governance research is still quite limited, with only 26 articles identified. According to the literature review, the research of inter-organisational IT governance has been published in nine conference papers and 17 journal articles since the start of this millennium. The journal that has published the most inter-organisational IT governance-related articles is the *European Journal of Information Systems* with three articles. There are 22 different persons mentioned as the first author of an article. Dr Acklesh Prasad is the most active author with two journal articles and one conference paper. Of the reviewed 26 articles, 15 focused on guidance or strengthening current understanding, three articles monitored implementation, four articles suggested improvements or extensions, and five articles focused on a concept, definition or model development. Empirical content was used in 19 articles, and seven articles used a conceptual approach.

The review of existing research revealed that research for inter-organisational IT governance is still lacking compared to research focusing on intra-organisational IT governance research, which creates many research opportunities for the future. Article I provides answers to RQ3 clarifying the current state of the inter-organisational IT governance research and confirming the possibility of using theories that explain intra-organisational IT cooperation in the inter-organisational context.

4.1.2 Article II

The article 'Formation of Voluntary Inter-Organizational IT Governance for Healthcare and Social Welfare IT - Theoretical Background and Empirical Evaluation' by Dahlberg and Helin (2014) examined the formation of IT governance between healthcare and social welfare organisations.

The results of the study indicate the importance of ICT cooperation. Despite the fears of trust in cooperation, the benefits of the ICT cooperation were valued highly. Participants were willing to start using IT governance principles for their ICT cooperation due to the known benefits. However, in some cases, the practical knowledge of IT governance among the participants was not sufficient to convince them about the benefits of ICT cooperation. Support from the theoretical background was needed to receive a complete understanding of the IT governance principles and benefits. The study confirmed the usability of RBV, TCE and the social network theory for increasing the IT governance knowledge in the inter-organisational ICT cooperation context. The results of the survey confirmed the importance and priorities of the IT governance principles and the benefits of ICT cooperation.

The study provides answers to RQ1 by clarifying the reasons for ICT cooperation and to RQ2 by confirming the importance of IT governance principles in creating the IT governance arrangement between organisations. The study also answers RQ3 by confirming the usability of RBV, TCE and Granovetter's (1973) social network theory in the inter-organisational context.

4.1.3 Article III

The article 'Volume, Benefits and Factors that Influence Inter-municipal ICT Cooperation in Relation to ICT-related Social Services and Healthcare Services' by Helin and Dahlberg (2017) examined the ICT cooperation of Finnish municipalities.

The research showed that municipal ICT cooperation and IT governance differ significantly in Finland. There are municipalities which do not cooperate with their nearby municipalities in ICT and other municipalities which cooperate actively and have joint IT governance. According to the results, the municipalities that run their IT through a well-organised ICT cooperation are capable of receiving significant benefits through their cooperation.

The study provides answers to RQ1 by highlighting the reasons and expectations that public organisations (municipalities) have about ICT cooperation, and the benefits the municipalities receive from the ICT cooperation. The study also answers RQ2 by emphasising the importance of well-structured and well-governed ICT cooperation.

4.1.4 Article IV

The article 'How IT Governance Practices Contribute to Inter-Municipal ICT Cooperation and Its benefits: Indeed, "The Emperor Has No Clothes' by Dahlberg and Helin (2017) examines how ICT cooperation between municipalities is carried out and governed in 20 municipal regions in Finland.

In this research, we found that municipal regions have different ways of working in IT. Several municipalities are not currently cooperating in ICT while some municipalities cooperate using various IT governance practices. Municipalities also receive different benefits from their cooperation.

The research showed that analysing secondary data provides a useful option for the research of IT governance and ICT cooperation. The research also shows how empirical studies can be conducted using secondary data as a methodological choice. The article confirms the role and importance of social ties in the establishment of ICT cooperation. Closer examination of Granovetter's (1973) social network theory also confirmed the importance of weak ties to the establishment of cooperation.

The study provides answers to RQ1 by identifying the reasons for ICT cooperation and the benefits that public organisations are looking forward to receiving from ICT cooperation. The study emphasises the importance of well-structured and well-governed ICT cooperation as focused on by RQ2. It extends the current cooperation research from intra-organisational contexts to inter-organisational contexts. By doing so, this research provides arguments for RQ3 by confirming the usability of selected theories for the inter-organisational context.

4.1.5 Article V

The article 'SECOs as a Mean to Survive? – Case Municipal ICT' by Helin (2017) extends the ICT cooperation research and clarifies whether the municipal ICT cooperation can be considered a software ecosystem (SECO). In a SECO, actors constituting a proper ecosystem correspond to the participants that can be identified in a municipal ICT cooperation.

In the research, the theoretical base is built on the RBV theory, TCE theory and Granovetter's (1973) social network theory. This theory base was used with definitions from the software environment. The empirical research was conducted among three municipal ICT cooperation arrangements (ecosystems) in Finland, and the data is based on interviews with leading experts in the ecosystems.

The study confirms the similar structures for municipal ICT cooperation and a SECO. The same actors, such as the hardware (HW) and the software (SW) platforms are in place, and the vendor companies are often the same. However, according to the definition of a SECO, a software provider should form the core of an ecosystem. However, this is not the case in municipal ICT cooperation. In the municipal context, a municipality, which is often a regionally leading municipality or a major city, acts as a leader in an inter-municipal ICT cooperation. Thus, ICT cooperation between municipalities does not fulfil the definition of a SECO (Manikas and Hansen, 2013). In the case of municipal ICT cooperation, a business ecosystem is a correct term to use (Moore, 1993).

The study clarifies the reasons for inter-municipal IT cooperation. The three most mentioned reasons were cost savings, improvements to IT efficiency and IT procurement/outsourcing. The study provides answers to RQ1 and offers further support for RQ2 by clarifying the structures and IT governance models of inter-municipal ICT cooperation and making a distinction between municipal ICT cooperation and SE-COs.

4.2 Results

In this thesis, the practical target is to examine ICT cooperation between municipalities, obtain a better understanding of the reasons for cooperation and to establish how municipalities are cooperating and governing the ICT cooperation. The thesis strives to determine whether the theories used in the intra-organisational ICT cooperation could be expanded to inter-organisational ICT cooperation. To fulfil the research objective, three RQs were set. The following paragraphs review all three RQs and discuss the results obtained in the articles.

As the municipalities in Finland vary in terms of the number of residents, industrial structures and locations, and their needs and expectations for inter-municipal ICT cooperation, RQ1 examines why Finnish municipalities cooperate in ICT. Articles II, III, IV and V examined this issue. RQ2 clarifies how inter-municipal ICT cooperation is organised and governed. Articles II, III and V investigate this topic. RQ3 aims to explain ICT activities between organisations using existing theories about the organisation's internal IT governance principles and IT governance benefits. Articles I, II and IV aim to clarify the topic and answer RQ3.

RQ1: What reasons do the municipalities have to participate in the inter-municipal ICT cooperation?

Article II examined the formation of inter-organisational IT cooperation and IT governance in a broad public environment consisting of ICT experts from healthcare, social welfare and municipal functions. Even though participants had only a little practical knowledge about IT governance, they valued the ICT cooperation. They were willing to start using IT governance principles for their ICT cooperation due to the possible benefits. The reasons to start ICT cooperation, including the need to pool, share and exchange IT resources.

Article III found that inter-municipal ICT cooperation was usually somewhat or very limited among the studied municipalities. Only three of the 20 regional municipal areas were identified to show noticeable and well-organised ICT cooperation. However, even within the other regional municipality areas, the attitude to ICT cooperation was favourable. While ICT cooperation was limited and restricted often only to joint purchases, the reasons for ICT cooperation were not explicitly specified. Expected benefits from ICT cooperation included economic and non-economic issues in which discussions and experience sharing with colleagues were highly valued.

Article IV revealed inter-municipal ICT cooperation to be very limited but showed that visions for better cooperation exist. Most of the reasons for ICT cooperation relate to cost savings and increased value creation. The main reason to improve ICT cooperation was seen as a means for cost savings. Otherwise, ICT cooperation was considered useless, and 7 out of 17 municipalities were planning to develop their ICT services alone. To receive potential benefits from ICT cooperation, input from the municipality's civil servants is needed. However, civil servants were often too busy, or they had no interest to provide input. They may even be missing the required competencies to contribute. Especially when civil servants believe that they receive no benefit from ICT services in their work; thus, they do not promote the inter-municipal ICT cooperation.

Article V found that all studied municipal groups had similar reasons for ICT cooperation. The most important economic reasons for cooperation were potential cost savings and resource optimisation while the most important non-economic reasons were communication between the ICT colleagues and the development of ICT skills and knowledge.

In summary, organisations were optimistic about the benefits available through ICT cooperation. The benefits, which are strongly expected, are non-economic and mainly related to personal communication, i.e. communication with ICT colleagues, new ICT skills and knowledge development but also resource optimisation is expected. In addition, ideas and willingness for ICT cooperation exist. Most of the economic reasons for potential ICT cooperation were related to cost savings and increased value creation. These are the reasons answering to RQ1.

RQ2: How is the inter-municipal ICT cooperation governed?

Article II examined the formation of inter-organisational ICT cooperation among public organisations. The proposed and implemented IT governance arrangement was based on two different governance bodies: steering operational IT governance work and an IT governance council. The essential services for organisations were identified and divided into three groups: (1) jointly governed and locally managed, (2) locally governed and managed and (3) TBD (to be decided). Accountabilities of the critical services/processes were determined. The study confirmed the importance of other constructs in addition to the IT governance body that is needed when establishing inter-organisational IT governance.

Article III identified inter-municipal ICT cooperation as still quite rare and restricted, often covering only joint ICT-related purchases. However, attitudes toward more extensive ICT cooperation are favourable. In the study, we identified only three of 17 regional areas which had well-organised ICT cooperation in place. In one regional municipal area, municipalities shared the ICT managerial function, and in another regional municipal area, they had a jointly owned ICT service company. In the third regional municipal area, municipalities had joint inter-organisational ICT governance, where they divided the ICT services into national, regional and municipal level activities and acted accordingly. All three regional municipal areas which had a well-organised ICT cooperation perceived significant benefits compared to the areas where the ICT cooperation was limited. In the areas where the ICT cooperation was limited, ICT experts were hoping for more active cooperation.

Article V examined the inter-municipal ICT cooperation between three municipality groups' organisations and IT governance structures. In principle, the municipality groups were similar based on the number of municipals in each group and the

number of inhabitants in the municipalities. However, the ICT cooperation was organised differently in each group. In one municipality group, ICT cooperation was voluntary without any official commitments. In the second group, the participating municipalities decided on the extent of their commitments, and the ICT cooperation was fulfilling the decisions. In the third municipality group, all participating municipalities were fully involved in ICT cooperation. Despite different targets of the ICT cooperation and different governance structures among the three groups, all municipality groups were satisfied with the cooperation and their joint achievements.

In summary, inter-municipal ICT cooperation has been implemented very differently in Finland. Based on the studies included in the thesis, some municipalities do not cooperate with ICT-related issues with their nearby municipalities, and other municipalities have well-established ICT cooperation. In addition, many municipalities conduct some form of ICT cooperation without any explicit reasons or targets. However, it can be seen that the municipalities that belong to a well-established, well-governed ICT cooperation receive benefits that the municipalities which do not participate fail to receive. Most of the municipalities have realised the benefits and thus favour the idea of starting ICT cooperation or joining an established ICT cooperation group. Thus, RQ2 has been answered.

RQ3: What are the possibilities to explain IT activities between organisations using existing theories about an organization's internal IT governance principles and IT governance benefits?

Article I identified the current state of the inter-organisational IT governance research. The study revealed that research for inter-organisational IT governance is still relatively limited, with only 26 articles published since 2000. However, the research confirmed that voluntary inter-organisational IT governance provides considerable benefits through ICT cooperation, where different competences are needed for successful cooperation. IT governance can use different organisational models, e.g. a federal governance structure. IT resources and management processes create synergies in inter-organisational cooperation. Organisations should sustain their IT capabilities, while management for IT resources may be a challenge in cooperation. The study revealed that theories (RBV, TCE and social network theory) used to describe IT cooperation in the intra-organisational context can be used in the inter-organisational IT governance context.

In Article II, the currently used constructs of the RBV, the TCE and Granovetter's (1973) social network theory provided support for understanding the interdisciplinary nature and theoretical background of the IT governance principles and benefits. This confirms that RBV, TCE and social network theory can enhance IT governance knowledge. The abovementioned theories also offer valid constructs to enhance IT

governance knowledge, and they represent the IT governance principles and benefits of ICT cooperation. The IT governance principles and benefits, which were planned using the enhanced IT governance knowledge, were important when creating the voluntary inter-organisational IT governance arrangement. This is important because current IT governance research has focused on investigating single organisation IT governance. The RBV and the TCE focus in a broader environment, which includes single organisations, markets and networks/alliances. RBV and TCE are also widely used in the IT outsourcing research, which makes the connection to the IT governance research promising.

Article IV found that inter-municipal ICT cooperation and the usage of IT governance practices were limited. In the municipality groups, where joint IT governance arrangements were agreed, the RBV and the TCE propositions were followed. The IT governance arrangements were the same in the intra-organisational as in the inter-organisational contexts. They followed the propositions of the IT governance practice research, using the same elements, namely structure, process, and relational mechanism practices. As the relational mechanism was the most used IT governance practice, this practice is proposed as the starting point when building ICT cooperation and IT governance between municipalities. As discussed in Granovetter's (1973) social network theory, the elements of trust and social ties are needed for ICT cooperation and IT governance. It was noted that effective IT governance created the trust and social ties between municipalities. Because of ICT cooperation between municipalities, economic and non-economic benefits were available; joint purchases and resource sharing were the most common benefits received. It became evident in the research that the usefulness of weak ties between municipality ICT experts was worsening due to intra-municipal strong ties. This finding highlighted the importance of intra-municipal issues, leaving the inter-municipal issues with lower importance and thus weakening the IT cooperation. The social network theory proved to be particularly useful in explaining the issues which affect the success of intermunicipal ICT cooperation.

Our research confirmed the ability to use theories for intra-organisational IT governance in the inter-organisational context. The same IT governance arrangements, structure, process and relational mechanism practices used in intra-organisational IT governance were also implemented in inter-organisational arrangements as proposed by the IT governance practice research. Trust and social ties are needed for intermunicipal ICT cooperation and IT governance, as described by Granovetter's (1973) social network theory. Our research also confirmed that more benefits are available to municipalities involved in ICT cooperation compared with municipalities which are not cooperating. RQ3 has, therefore, been answered.

The findings from the articles show the overall status for municipal ICT cooperation in Finland with the reasons the municipalities have for cooperation and the

forms of IT governance that the municipalities have selected to use. The usefulness of current theories used in intra-organisational ICT cooperation was shown to be relevant in an inter-organisational context. Based on the research conducted and the articles included in the thesis, it can be said that the objective of the thesis has been fulfilled.

5 Implications and Conclusions

In this chapter, I discuss the implications of the thesis and the theoretical and practical contributions of the thesis. To conclude, I present a summary of the thesis and thoughts and suggestions for future research.

5.1 Implications

According to the research conducted, ICT is still often seen as a function that needs little attention and few resources. Many Finnish municipalities execute ICT independently with few resources, and they fail to see any added value in ICT cooperation. However, the municipalities which have understood the importance of ICT in general and in joint ICT in particular, are participating in well-governed ICT cooperation systematically with their nearby municipalities and they are receiving the benefits. Although the structure of well-governed ICT cooperation likely differs, municipalities should aim to be fully involved in ICT cooperation instead of working in isolation. However, the reasons for and benefits from inter-municipal ICT cooperation vary according to the prevailing economic and political situation.

Additionally, depending on the decisions by the national government, the reasons and benefits for the inter-municipal cooperation may change overnight. Further, economic realities should drive municipalities towards closer ICT cooperation. Also, as predicted by Gil-Garcia (Gil-Garcia, 2012), ICT cooperation between government organizations and local government is intensifying and is still in demand.

In our research, the importance of social relationships has been highlighted, strengthening the role of trust in inter-organizational cooperation as also discussed earlier (Pardo, Gil-Garcia & Burke, 2006). Based on our findings, the models of municipal ICT cooperation vary a lot in Finland. The Finnish models of inter-municipal ICT cooperation are closer to the Swedish model (Juell-Skielse et al., 2017) than the Italian model (Ferro & Sorrentino, 2010) because Finnish municipalities are generally looking for some benefits through the ICT cooperation. Municipalities in Finland are using the cooperation modes of the Convention, the Framework agreement and the Consortium. Unlike in Sweden, the Limited company model is also

used in Finland (e.g. the model of cooperation in North Karelia). The New public body form of collaboration was not identified in Finland.

Overall, it can be said that in Finland, inter-municipal ICT cooperation occurs in many different ways. There are municipalities, which try to get some benefits like improved efficiency or shared services through ICT cooperation, and there are municipalities, which are doing ICT cooperation to save money through a joint procurement. Also, there are municipalities, which have made a legal agreement about ICT cooperation with roles and responsibilities, and there are municipalities, which have established a joint company to take care of their ICT issues.

For scientific research, our research contributes to the still limited research on inter-municipal ICT cooperation by studying the current state of Finnish inter-municipal cooperation. The possibility of using different models of ICT cooperation in various cases of municipal cooperation gives freedom to municipal decision-makers and does not limit their choice. It is up to decision-makers to choose the guiding factors, the purpose of the collaboration and the desired benefits.

By highlighting the necessity and importance of social networks, our research emphasizes the importance of personal relationships in operation and development of municipal ICT functions. By validating the usability of existing intra-organizational theories in an inter-organizational environment, our research extends the potential of current theories in future research.

5.2 Theoretical Contribution

In Article I, we conducted a literature review about the current state of inter-organisational IT governance research. Based on the review, we identified a research gap where intra-organisational IT governance was dominating research, and inter-organisational IT governance was mostly absent. The RBV, the TCE, and social network theories have been widely used in the governance arrangement research, and they provide several elements for measuring cost savings and value increases. These theories have also promoted performance benefits. Based on our research, measurable benefits are also available through inter-organisational cooperation, if IT governance arrangements are systematic and well organised. Organisations can implement the same research-proposed components used in intra-organisational arrangements such as the structure, process and relational mechanism practices in the inter-organisational arrangements. Also, these theories highlight the benefits of inter-organisational cooperation and describe IT governance principles. The role of trust and social ties are vital for cooperation between municipalities as well for the joint IT governance as discovered in our

research. These findings underline the importance of Granovetter's (1973) social network theory in inter-organisational cooperation.

Our research confirms that organisations can use RBV, TCE, and the social network theory to extend IT governance research from current intra-organisational research to a more topical inter-organisational context providing a theoretical contribution. The role of RBV and TCE focuses more on cost savings and value increases, meaning that organisations do not have enough resources (money or people) to do everything by themselves, and they have to cooperate. According to our research, the role of the social network theory is crucial in making cooperation work efficiently and productively. This finding adds a new angle to the theoretical approach.

5.3 Practical Contributions

Our research makes several practical contributions. Based on our research, the importance of social connections cannot be underestimated. Municipal IT cooperation often starts with informal meetings where work colleagues from different municipalities discuss professional issues. Informal meetings may progress to a more systematic and well-governed cooperation that may ultimately have real measurable benefits. This outcome is why practitioners are encouraged to focus on the social connections of inter-organisational IT governance. The role of social connections and trust is vital in the formation of IT governance between organisations. Because the role of social connections and trust is so essential, it would make sense to encourage and allow people from different organisations to organise official or unofficial events or the like to meet new people, get to know them better and build trust to them. Especially in the municipal sector in Finland, where the employment relationships tend to be long, this would make all forms of cooperation much more accessible. With the help of new contacts, municipal civil servants can learn new things and discuss their challenges even without closer cooperation.

Also, practitioners should identify real cost savings and value increase possibilities, not forgetting the importance of sufficient resources for implementation. They should also take into consideration IT governance principles and IT cooperation benefits when they are planning new IT governance arrangements. When joint decisions form the basis for the IT cooperation between organisations, when there is sufficient amount of resources (both money and human) available and when the IT governance is well established, the participants will be satisfied. Organisations can expect better ICT services and other measurable benefits. However, even though inter-municipal cooperation is welcomed, official involvement in forms of various incentives should be planned carefully, while the benefits may be uncertain (Sorrentino and Simonetta,

2013). As such, inter-organisational cooperation can be a solution for many municipal challenges. However, despite its promise, inter-organizational cooperation as such is not a guarantee of success (Ferro and Sorrentino, 2010).

5.4 Limitations

Even though the current thesis has attempted to obtain a broad understanding of the status of inter-municipal ICT cooperation, there are some limitations. We conducted all the research included in the thesis in one country and during a relatively short period. The fact that economic and political realities are varying may change the reasons and benefits available from ICT cooperation even in Finland. Future research conducted in another country could bring different results due to cultural, economic, political or other reasons.

A limitation is that we were not able to identify much research regarding the intermunicipal ICT cooperation focus. A reason for this may be the fact that different countries use different vocabulary when they are discussing ICT issues of their local government. Targeting to other research areas and using different search criteria, other research concerning inter-municipal ICT cooperation may exist.

A limited number of public organisations, mainly municipalities but also healthcare and social welfare units which have been at the core of our research, may cause some biases due to geographic, financial or structural reasons. Also, the limited number of interviews and the selection of ICT experts who participated in the interviews may affect the generalisability of the results.

The results concerning ICT cooperation and the IT governance benefits of ICT cooperation are clear and consistent. Still, the number of municipalities which were found to have well-organised inter-municipal ICT cooperation and joint IT governance is limited. This small number may limit the comparison of the achievements of ICT cooperation for different municipality groups, i.e. one with organised IT governance vs another with non-organised IT governance.

The main theoretical limitation is that the theoretical basis used throughout the thesis is rather narrow. Only a few fundamental theories have been used throughout the research. Other theories and frameworks, such as a knowledge-based theory of inter-firm collaboration (Grant and Baden-Fuller, 1995) and a dynamic capabilities framework (Teece et al., 1997), exist. These theories also focus on cooperation between organisations, and the benefits of such collaboration. Extending the theoretical basis for these theories may bring new observations and conclusions.

5.5 Future Research

Our research provides a sound basis for future IT governance and ICT cooperation research where RBV, TCE and social network theories enable a robust and widely used basis. Researchers use RBV to explain the limited amount of resources in an organisation, and they use TCE to explain the distribution of tasks and challenges among the participants. In our research, we brought the importance of social network theory into the research of inter-organisational cooperation. Future research needs to evaluate the impact of social ties while implementing IT governance.

While research for inter-organisational ICT cooperation at the municipal or local government level is still relatively rare, more studies will be needed. Inter-municipal ICT cooperation research in a few countries (e.g. Italy, Nordic countries, USA) has established a sound foundation and research should be continued. It would be interesting to compare the effectiveness of municipal cooperation and the mode of operations in different countries and share best practises. Since cooperation models vary a lot of Finland, it would be interesting to propose a similar approach to other countries. At the same, it could be useful for Finnish municipalities to better understand the cooperation models, which, e.g. Sweden or Norway, are using and further develop the Finnish approach. Cooperation between countries is welcome.

However, even a repetition of the current research conducted in the country of the current research (Finland) would be of significant value due to the new political situation. Because it usually takes time to build trust and create ICT cooperation and joint IT governance between organisations, longitudinal studies would offer an excellent research ground. While existing research has found significant advantages to inter-organisational ICT cooperation, future research could examine the concrete benefits received from ICT by planning the benefits in the ICT project planning and including a follow-up of the benefits at different phases of the project. Doing so would raise the understanding of the importance of ICT and could eliminate the lack of resources.

While IT governance research has focused mainly on single organisation governance and ignored inter-organisational cooperation, our research provides a good opening for future inter-organisational and market governance research. An example of market governance research is IT outsourcing research. Here, as RBV and TCE are also widely used, the connection to IT governance research exists. Another future research area could be supplier-customer networks and IT governance arrangements in those networks. Also, networks with a principal contractor and several subcontractors would be worth researching. In general, due to the current network-based business environment, organisations need increased IT governance and ICT cooperation research in a networked business environment.

In our research, we also observed that no municipality cooperation exists in the areas of data security and IT risk management, which invites conducting research on ICT cooperation and IT governance in these areas. The area of inter-organisational IT governance in health information governance also invites more research.

5.6 Conclusions

The thesis has provided an exciting journey into the ICT environment of public organisations, especially in the ICT environment of municipalities in Finland. It has been fruitful to notice that the role and appreciation of ICT are changing. More and more municipalities have realised the importance of ICT. For them, ICT is not just a necessary tool for emailing and bookkeeping, but it is a way of improving the living environment for their residents. They have also realised that for effective ICT operations, they need inter-organisational cooperation between municipalities. However, there is an issue which may cause municipalities to avoid inter-organisational ICT cooperation: the fear of a hostile municipal takeover. Small municipalities are afraid of a situation in which their more prominent municipal neighbour may grow its influence and eventually force its smaller neighbours to unite. However, in the cases of the current thesis, there were no signs or comments about this type of hidden agenda.

In Finland, the previous government's decision regarding a plan for a change in nationwide administration structures combined with a change in social welfare and healthcare structures increased attention toward the ICT issues. The government has changed, and the political intent is different. The planned change of the previous government will not take place. Still, authorities have noted the importance of municipal ICT-related issues and the development for public ICT is progressing within Finland. Some municipalities, especially the smaller ones, would like to have national guidelines regarding how local ICT should be developed, implemented and managed. However, such practical guidelines do not yet exist, and municipalities have to develop and implement their ICT alone. Due to limited resources, municipal ICT development activities often shine out of absence, and municipalities focus all efforts on daily ICT operations, alongside one's work. Municipal decision-makers should have a better understanding of the possibilities and about the potential ICT has to create better and more effective services for residents of municipalities. They should also have better visions about ICT as an enabler for the differentiation of municipalities.

There have been some comments about the structure, where one could divide public sector ICT activities into three levels: nationwide, region-wide and local level. Although it would make sense to do some ICT tasks at the local level, it would be

better to do some issues jointly on a regional level or with nearby municipalities. Local authorities could handle small, daily ICT operations and municipal specific matters at the local level. In contrast, authorities at a regional level could handle issues which have an effect on all municipalities in a region. Authorities at the national level could take care of generic applications and guidelines and provide general support for all municipalities. Further, municipalities and other public organisations should direct more focus on the networked business environment. Otherwise, all ICT operations will stay on normal daily mode, and no progress will take place.

Overall, municipal ICT will need more resources to ensure good results, prompt service development and digitalisation of services. The role and importance of interorganisational ICT cooperation among municipalities will grow. Thus, essential issues are how to organise ICT cooperation between municipalities, how to manage the governance of ICT cooperation permanently and how to organise cooperation between the municipal, regional and national levels. According to our research, there is not just a single solution for every situation. Various ways of how to organise ICT cooperation can be implemented, which may provide satisfactory results. However, municipalities with well-established ICT cooperation can expect to receive measurable benefits through their collaboration. ICT cooperation improves the efficiency of organisations, although the reasons for cooperation and the details of the benefits may change due to political or other reasons. In any case, municipalities with poor ICT operations, inadequate online services and weak digitalisation capabilities will not survive in the future.

List of References

- Abu-Musa, A. A. (2009). Exploring COBIT processes for ITG in Saudi Organizations: An empirical study. *The International Journal of Digital Accounting Research*, *9*, 99-126.
- Allen, I. E., & Seaman, C. A. (2007, July). Likert scales and data analyses. *Quality Progress*. Retrieved from http://rube.asq.org/quality-progress/2007/07/statistics/likert-scales-and-data-analyses.html
- Ali, S., & Green, P. (2007). IT governance mechanisms in public sector organisations: An Australian context. *Journal of Global Information Management*, 15(4), 41-63.
- Ali, S., & Green, P. (2008). IT governance mechanisms in public sector organisations: An Australian context. In G. M. Hunter (Ed.), *Handbook of research on information management and the global landscape* (pp. 458-473). IGI Publishing Hershey, PA.
- Ali, S., & Green, P. (2012). Effective information technology (IT) governance mechanisms: An IT outsourcing perspective. *Information Systems Frontiers*, 14(2), 179-193.
- Almeida, R., Pereira, R., & da Silva, M. M. (2013). IT governance mechanisms: A literature review. In J. Falcao e Cunha, M. Snene, & H. Novoa (Eds.). *Lecture notes in business information processing* (pp. 186-199). Springer Nature Switzerland AG.
- Amali, L. N., Mahmuddin, M., & Ahmad, M. (2014). Information technology governance framework in the public sector organizations. *Telkomnika (Telecommunication Computing Electronics and Control)*, 12(2), 429-436.
- Anica-Popa, I. (2012). Standards in IT governance. Quality Access to Success, 13(130), 110-112.
- The Association of Finnish Local and Regional Authorities. (2018). Finnish municipalities and regions. Retrieved from https://www.localfinland.fi/expert-services/finnish-municipalities-and-regions
- Atlas.ti. (n.d.). Qualitative data analysis. Retrieved from https://atlasti.com
- Barney, J. (1991). Firm resources and sustained competitive advantage. *Journal of Management*, 17(1), 99-120.
- Bel, G., & Warner, M. (2015). Inter-municipal cooperation and costs: expectations and evidence. *Public Administration*, 93(1), 52-67.
- Benbasat, I, Goldstein, D. K., & Mead, M. (1987). The case research strategy in studies of information systems. *MIS Quarterly*, 11(3), 369-386.
- Bin-Abbas, H., & Bakry, S. H. (2014). Assessment of IT governance in organizations: A simple intengrated approach. *Computers in Human Behavior*, 32, 261-267.
- Borgatti, S. P., & Foster, P. C. (2003). The network paradigm in organizational research: A review of typology. *Journal of Management*, 29(6), 991-1013.
- Bradley, R. V., Byrd, T. A., Pridmore, J. L., Thrasher, E., Pratt, R. M. E., & Mbarika, V. W. A. (2012). An empirical examination of antecedents and consequences of IT governance in US hospitals. *Journal of Information Technology*, 27(2), 156-177.
- Brown, C. V., & Magill, S. L. (1994). Alignment of the IS functions with the enterprise: Toward a model of antecedents. MIS Quarterly, 18(4), 371-403.
- Bögenhold, D. (2013). Social network analysis and the sociology of economics: Filling a blind spot with the idea of social embeddedness. *American Journal of Economics Sociology*, 72(2), 293-318.

- Chi, M., Zhao, J., George, J. F., Li, Y., & Zhai, S. (2017). The influence of inter-firm IT governance strategies on relational performance: The moderation effect of information technology ambidexterity. *International Journal of Information Management*, 37(2), 43-53.
- Chua, W. F. (1986). Radical developments in accounting thought. *The Accounting Review*, 61(4), 601-632.
- COBIT 5. (n.d.). COBIT 5: A business framework for the governance and management of enterprise IT. Manual. Retrieved from https://www.isaca.org/cobit/pages/default.aspx
- Coase, R. H. (1937). The nature of the firm. Economica, 4(new series), 385-405.
- Croteau, A-M., & Bergeron, F. (2009). Interorganizational governance of information technology. Proceedings of the 42nd Annual Hawaii International Conference on System Sciences (HICSS'09), 1-8
- Croteau, A-M., Bergeron, F., & Dubsky, J. (2013). Contractual and consensual profiles for an interorganizational governance of information technology. *International Business Research*, 6(9), 30-43.
- Dahlberg, T. (2016). The creation of inter-organisational it governance for social welfare and healthcare IT Lessons from a case study. *Internal Journal of Networking and Virtual Organisations*, 16(1), 38-71.
- Dahlberg, T., & Helin, A. (2014). Formation of voluntary inter-organizational IT governance for healthcare and social welfare IT - Theoretical background and empirical evaluation. *Proceedings* of the 8th International Conference on Theory and Practice of Electronic Governance (ICEGOV). Guimaraes, Portugal, October 27 - 30, 2014, pp. 320-329. ACM New York, NY, USA.
- Dahlberg, T., & Helin, A. (2017). How IT governance practices contribute to inter-municipal ICT cooperation and its benefits: Indeed, "the emperor has no clothes". *International Journal of IT/Business Alignment and Governance (IJITBAG)*, 8(2), 62-79.
- Dahlberg, T., & Helin, A. (2017). Volume, benefits and factors that influence inter-municipal ICT cooperation in relation to ICT-related social services and healthcare services. *Finnish Journal of eHealth and eWelfare (FinJeHeW)*, 9(4), 299-312.
- Dahlberg, T., & Kivijärvi, H. (2006). An integrated framework for IT governance and the development and validation of an assessment instrument. In HICSS '06: Proceedings of the 39th Annual Hawaii International Conference on System Sciences (HICSS'06), Kauia, HI, USA, IEEE, doi: 10.1109/HICSS.2006.57.
- Dale, B. G., Elkjaer, M. B. F., van der Wiele, A., & Williams, A. R. T. (2001). Fad, fashion and fit: An examination of quality circles, business process re-engineering and statistical process control. *International Journal of Production Economics*, 73(2), 137-152.
- Das, T.K., & Teng, B-S. (2000). A resource-based theory of strategic alliances. *Journal of Management*, 26(1), 31-61.
- Dawson, G. S., Denford, J. S., Williams, C. K., Preston, D., & Desouza, K. C. (2016). An examination of effective IT governance in the public sector using the legal view of agency theory. *Journal of Management Information Systems*, 33(4), 1180-1208.
- De Haes, S., & Van Grembergen, W. (2009). An exploratory study into IT governance implementations and its impact on business/IT alignment. *Information Systems Management*, 26(2), 123-137.
- De Haes, S., & Van Grembergen, W. (2013). Improving enterprise governance of IT in a major airline: A teaching case. *Journal of Information Technology Teaching Cases*, 3(2), 60-69.
- De Haes, S., & Van Grembergen, W. (2013). COBIT 5 and enterprise governance of information technology: Building blocks and research opportunities. *The Journal of Information Systems*, 27(1), 307-324.
- Ebers, M. (1997). Explaining inter-organizational network formation. In M. Ebers (Eds.). *The formation of inter-organizational networks*. New York, US: Oxford University Press. doi:10.1.1.111.5848.
- Eisenhardt, K. M. (1989). Building theories from case study research. *Academy of Management Review*, 14(4), 532-550.

- EUR-Lex (2003). The European Parliament. Regulation (EC) No 1059/2003 of the European Parliament and of the Council of 26 May 2003 on the establishment of a common classification of territorial units for statistics (NUTS). Retrieved from https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=uriserv:OJ.L_.2003.154.01.0001.01.ENG&toc=OJ:L:2003:154:TOC
- Ferro, E., & Sorrentino, M. (2010). Can intermunicipal collaboration help the diffusion of E-Government in peripheral areas? Evidence from Italy. *Government Information Quarterly*, 27, 17-25.
- Flak, L. S., Dertz, W., Jansen, A., Krogstie, J., Spjelkavik, I., & Olnes, S. (2009). What is the value of e Government and how can we actually realize it? *Transforming Government: People, Process and Policy*, 3(3), 220-226.
- Flak, L. S. & Solli-Saether, H. (2013). Benefits realization in e-government: Institutional entrepreneurship or just hype? In R. Sprague Jr. (Ed.): *Proceedings of 46th Hawaii International Conference on System Sciences*, 2062-2071.
- Gebre-Mariam, M., & Fruijtier, E. (2018). Countering the "dam effect": The case for architecture and governance in developing country health information systems. *Information Technology for Development*, 24(2), 333-358.
- Gereffi, G., Humphrey, J., & Sturgeon, T. (2005). The governance of global value chains. *Review of International Political Economy*, 12(1), 78-104.
- Geyskens, I., Steenkamp, J-B. E. M., & Kumar, N. (2006). Make, buy, or ally: A transaction cost theory meta-analysis. *Academy of Management Journal*, 49(3), 519-543.
- Gheorghe, M. (2010). Audit methodology for IT governance. Informatica Economica, 14(1), 32-42.
- Gheorghe, M. (2011). Risk Management in IT Governance Framework. *Economia: Seria Management*, 14(2), 545-552.
- Gil-Garcia, J. R. (2012). Towards a smart state? Inter-agency collaboration, information integration, and beyond. *Information Polity*, 17 (3,4), 269-280.Goldkuhl, G. (2012). Pragmatism vs interpretivism in qualitative information systems research. *European Journal of Information Systems*, 21, 135-146.
- Granovetter, M. S. (1973). The strength of weak ties. *American Journal of Sociology*, 78(6), 1360-1380.
- Granovetter, M. S. (1983). The strength of weak ties: A network theory revisited. *Sociological Theory*, 1, 201-233.
- Granovetter, M. S. (1985). Economic action and social structure: The problem of embeddedness. *American Journal of Sociology*, 91(3), 481-510.
- Granovetter, M. S. (2005). The impact of social structure on economic outcomes. *Journal of Economic Perspectives*, 19(1), 33-50.
- Grant, R. M., & Baden-Fuller, C. (1995). A knowledge-based theory of inter-firm collaboration. *Academy of Management Proceedings*, 1995(1), 17-21.
- Guha, J. & Chakrabarti, B. (2014). Making e-government work: Adopting the network approach. *Government Information Quarterly*, *31*, 327-336.Gulati, R. (1999). Network location and learning: the influence of network resources and firm capabilities on alliance formation. *Strategic Management Journal*, *20*(5), 397-420.
- Heindrickson, G., & Santos, C. D. (2014). Information technology governance in public organizations: How perceived effectiveness relates to three classical mechanisms. *Journal of Information Systems and Technology Management*, 11(2), 297-326.
- Helin, A. (2017). SECOs as a mean to survive? Case municipal ICT. Proceedings of the 9th International Workshop on Software Ecosystem (IWSECO 2017). Publication of Conference IWSECO 2017 9th International Workshop on Software Ecosystem. Espoo, Finland, November 29, 2017.
- Helin, A. (2019). Inter-organizational IT governance research A literature review. *International Journal of IT/Business Alignment and Governance (IJITBAG)*, 10(1).

- Henderson, J. C., & Venkatraman, H. (1999). Strategic alignment: Leveraging information technology for transforming organizations. *IBM Systems Journal*, 38(2/3), 472-484.
- Héroux, S., & Fortin, A. (2013). The internal audit function in information technology governance: A holistic perspective. *Journal of Information Systems*, 27(1), 189-217.
- Héroux, S., & Fortin, A. (2018). The moderating role of IT-business alignment in the relationship between IT governance, IT competence, and innovation. *Information Systems Management*, 35(2), 98-123.
- Hophmayer-Tokich, S., & Kliot, N. (2008). Inter-municipal cooperation for wastewater treatment: Case studies from Israel. *Journal of Environmental Management*, 86, 554-565.
- Hyrynsalmi, S. (2014). Letters from the war of ecosystems An analysis of in-dependent software vendors in mobile application marketplaces. *TUCS Dissertations* No 188, December, 38-41 Juvenes Print, Turku, Finland.
- Hyvärinen, S., & Parviainen, J. (2018). Kuntien tietotekniikkakartoitus 2018 (*The municipal ICT survey 2018*). Kuntaliitto (Association of Finnish local and regional Authorities), Hki, 12-13. Retrieved https://www.kuntaliitto.fi/sites/default/files/media/file/Tietotekniikkakartoitus 2018_SU%20RU.pdf.
- Hyvönen, H. (2015). Alignment and leadership in ICT strategies of private and public organizations. Ph.D. Dissertation. University of Jyväskylä, Faculty of Information Technology, Jyväskylä University Printing House, Jyväskylä.
- Hyvönen, K. (2015). ePS2, Tietohallintohteistyö (ePS2, IM Cooperation, Finnish-language material). Paper presented at the SeutuICT conference: 4th November 2015; Turku, Finland.
- Iden, J., & Eikebrokk, T. R. (2014). Using the ITIL process reference model for realizing IT Governance: An empirical investigation. *Information Systems Management*, 31(1), 37-58.
- Iliescu, F. (2010). Auditing IT Governance. *Informatica Economica*, 14(1), 93-102.
- ISO/IEC 35800:2015 (2015). International Organization for standardization Corporate Governance of information technology. International Standard (IS). Retrieved from https://www.iso.org/standard/62816.html.
- ITIL (2018). The home page of ITIL (abbreviation for Information Technology Infrastructure Library). Retrieved from https://www.axelos.com/best-practice-solutions/itil/what-is-itil.
- Jansen, S., & Cusumano, M. A. (2013). Defining software ecosystems: A survey of software platforms and business network governance. In S. Jansen, S. Brinkkemper, & M. A. Cusumano (Eds.). Software ecosystems: Analyzing and managing business networks in the software industry (pp. 13-28). Northampton, MA: Edward Elgar Publishing.
- Juell-Skielse, G., Lönn, C-M., & Päivärinta, T. (2017). Modes of collaboration and expected benefits of inter-organizational E-government initiatives: A multi-case study. *Government Information Quarterly*, 34(4), 578-590.
- Klein, H. K, & Myers, M. D. (1999). A Set of Principles for Conducting and Evaluating Interpretive Field Studies in Information Systems. *MIS Quarterly*, Vol. 23 No.1, 67-94.
- Kolb, D. A. (1984). Experiential learning: Experience as the source of learning and development. Englewood Cliffs, NJ: Prentice-Hall.
- Kvale, S. (2006). Dominance through interviews and dialogues. *Qualitative Inquiry*, 12(3), 480-500.
- Köbler, F., Fähling, J., Krcmar, H., & Leimeister, J. M. (2010). IT governance and types of it decision makers in German hospitals. *Business & Information Systems Engineering*, 2(6), 359-370.
- Lagsten, J., & Nordström, M. (2015). Evaluating an IT governance model-in-use. *Systems, Signs & Actions*, 9(1), 93-112.
- Layne, K., & Lee, J. (2011). Developing fully functional E-government: A four stage model. Government Information Quarterly, 18(2), 112-136.
- Leclercq-Vandelannoitte, A. (2018). From sovereign IT governance to liberal IT governmentality? A Foucauldian analogy. *European Journal of Information Systems*, 27(3), 326-346.
- Loh, L. & Venkatraman, N. (1992). Diffusion of information technology outsourcing: Influence sources and the Kodak effect. *Information Systems Research*, *3*(4), 334-358.

- Lombardi, R., Del Giudice, M., Caputo, A., Evangelista, F., & Russo, G. (2016). Governance and assessment insights in information technology: The Val IT model. *Journal of the Knowledge Economy*, 7(1), 292-308.
- Lorences, P. P., & Avila, L. F. G. (2013). The evaluation and improvement of IT governance. *Journal of Information Systems and Technology Management*, 10(2), 219-234.
- Lunardi, G. L., Becker, J. L., Macada, A. C. G., & Dolci, P. C. (2014). The impact of adopting IT governance on financial performance: An empirical analysis among Brazilian firms. *International Journal of Accounting Information Systems*, 15(1), 66-81.
- McCaffrey, D., Faerman, S., & Hart, D. (1995). The appeal and difficulties of participative systems, *Organization Science*, 6(6), 603-627.
- Melville, N., Kraemer, K., & Gurbaxani, V. (2004). Review: Information technology and organizational performance: An integrative model of IT business value. *MIS Quarterly*, 28(2), 283-322.
- Magni, M., Angst, C. M., & Agarval, R. (2012-2013). Everybody needs somebody: The influence of team network structure on information technology use. *Journal of Management Information Systems*, 29(3), 9-42.
- Manikas, K. & Hansen, K. M. (2013). Software ecosystems A systematic literature review. *The Journal of Systems and Software*, 86(5), 1294-1306.
- Mason, J. (2006). Mixing methods in a qualitatively driven way. Qualitative Research, 6(1), 9-25.
- Melville, N., Kraemer, K., & Gurbaxani, V. (2004). Information technology and organizational performance: An integrative model of IT business value. *MIS Quarterly*, 28(2), 283-322.
- Mingers, J. (2001). Combining IS research methods: Towards a pluralist methodology. *Information Systems Research*, 12(3), 240-259.
- Ministry of Finance, (2015). ICT-muutostuki (The ICT-change support; Finnish-language programme material). Helsinki: JulkICT-toiminto. Retrieved from https://wiki.julkict.fi/julkict/projektit/Kuntauudistus/ict-muutostuki-1
- Ministry of Finance (2018a). Number of Municipalities. Retrieved from http://vm.fi/kuntien-lukumaara?p_p_id=56_INSTANCE_SSKDNE5ODInk&p_p_lifecycle=0&p_p_state=normal&p_p_mode=view&p_p_col_id=column-2&p_p_col_count=1&_56_INSTANCE_SSKDNE5ODInk_languageId=fi_FI
- Ministry of Finance (2018b). Regional reform. Retrieved from http://alueuudistus.fi/en/general-information-reform.
- Ministry of Finance (2019a). Uusi sote-ja maakuntarakenne (The new social and health care and regional structure). Retrieved from https://alueuuditus.fi/ toimijat-uudessa-maakunta-ja-sote-rakenteessa-kuva-tekstina
- Ministry of Finance (2019b). Health and social services. Retrieved from https://alueuudistus.fi/palvelut-ja-valinnanvapaus?
- Ministry of Justice (1999). The constitution of Finland. Retrieved from https://www.finlex.fi/en/laki/kaannokset/ 1999/en19990731.pdf, 24-25.
- Moon, M. J. (2002). The evolution of e-government among municipalities: Rhetoric or reality? *Public Administration Review*, 62(4), 424-433.
- Moore, J. F. (1993). Predators and prey: A new ecology of competition. *Harvard Business Review*, 71(3), 75-86.
- Myers, M. D., & Newman, M. (2007). The qualitative interview in IS research: Examining the craft. *Information and Organization*, 17(1), 2-26.
- Nicho, M., & Al Mourad, B. (2012). Success factors for integrated ITIL deployment: An IT governance classification. *Journal of Information Technology Case and Application Research*, 14(1), 25-54.
- Newsted, P. R., Huff, S. L., & Munro, M. C. (1998). Survey instruments in information systems. *MIS Quarterly*, 22(4), 553-553.
- Nfuka, E. N., & Rusu, L. (2013). Critical success framework for implementing effective IT governance in Tanzanian public sector organizations. *Journal of Global Information Technology Management*, 16(3), 53-77.

- O'Leary, R., & Vij, N. (2012). Collaborative public management: Where have we been and where are we going? *The American Review of Public Administration*, 42(5), 507-522.
- Orlikowski, W. J., & Baroudi J. J. (1991). Studying information technology in organizations: Research approaches and assumptions. *Information Systems Research*, 2(1), 1-28.
- Osborn, R., & Baughn, C. C. (1990). Forms of interorganizational governance for multinational alliances. *Academy of Management Journal*, 33(3), 503-519.
- Oxford Reference (2019). A dictionary of business and management (6 ed.) Retrieved from http://www.oxfordreference.com.ezproxy.utu.fi/view/10.1093/acref/9780199684984.001.0001/acref-9780199684984-e-5186?rskey= nQnc4c&result=1
- Pang, M. (2014). IT governance and business value in the public sector organizations The role of elected representatives in IT governance and its impact on IT value in U.S. state governments. *Decision Support Systems*, 59(1), 274-285.
- Pardo T. A., Gil-Garcia, J. R., & Burke, G. (2006). Building Response Capacity through Cross-boundary information sharing: The critical role of trust. *Publication of eChallenges conference*, Barcelona, Spain.
- Park, N. K., Mezias, J. M., & Song, J. (2004). A resource-based view of strategic alliances and firm value in the electronic marketplace. *Journal of Management*, 30(1), 7-27.
- Pereira, R., & da Silva, M. M. (2012). Towards an integrated IT governance and IT management Framework. *Proceedings of the IEEE 16th International Enterprise Distributed Object Computing Conference (EDOC)*, September 10-14, IEEE, 191-200, Beijing, China.
- Penrose, E. T. (1959). The theory of the growth of the firm. Oxford, UK: Oxford University Press.
- Peterson, R. (2006). Crafting information technology governance. *Information Systems Management*, 21(4), 7-22.
- Pfeffer, J., & Salancik, G. R. (1978). The external control of organizations: A resource dependence perspective. New York, NY: Harper and Row.
- Pinsonneault, A., & Kraemer, K. L. (1993). Survey research methodology in management information systems: An assessment. *Journal of Management Information Systems*, 10(2), 75-105.
- Polkinghorne, D. E. (2005). Language and meaning: Data collection in qualitative research. *Journal of Counselling Psychology*, 52(2), 137-145.
- Population Register Centre Finland (2018). (In Finnish: Väestörekisterikeskus VRK). The number of local residents. Retrieved from http://vrk.fi/default.aspx?docid=8816&site=3&id=0
- Prasad, A., Heales, J., & Green, P. (2010). A capabilities-based approach to obtaining a deeper understanding of information technology governance effectiveness: Evidence from IT steering committees. *International Journal of Accounting Information Systems*, 11(3), 214-232.
- Prasad, A., Green, P., & Heales, J. (2012). On IT governance structures and their effectiveness in collaborative organizational structures. *International Journal of Accounting Information Systems*, 13(3), 199-220.
- Päivärinta T., & Dertz W. (2008). Pre-determinants of implementing IT benefits management in Norwegian municipalities: Cultivate the context. In M. A. Wimmer, H. J. Scholl, & E. Ferro (Eds.). *Electronic Government. EGOV 2008. Lecture Notes in Computer Science, vol 5184*. Springer, Berlin, Heidelberg.
- Rayle, L., & Zegras, C. (2013). The Emergence of Inter-Municipal Collaboration: Evidence from Metropolitan Planning in Portugal. *European Planning Studies*, 21(6), 867-889.
- Schultze, U., & Avital, M. (2011). Designing interviews to generate rich data for information systems research. *Information and Organization*, 21, 1-16.
- Sireci, S. G. (1998). The construct of content validity. *Social Indicators Research*, November, 45(1–3), 83-117.
- Smith, A. L., Bradley, R. V., Bichescu, B. C., & Tremblay, M. C. (2013). IT governance characteristics, electronic medical records sophistication, and financial performance in U.S. hospitals: An empirical investigation. *Decision Sciences*, 44 (3), 483-516.

- Sohal, A. S., & Fitzpatrick, P. (2002). IT governance and management in large Australian organisations. *International Journal of Production Economics*, 75, 97-112.
- Sorrentino, M., & Ferro, E. (2008). Does the answer to e-government lie in intermunicipal collaboration? An exploratory Italian case study. In *M. Wimmer, H.J. Scholl, E. Ferro (Eds.), Electronic government: Proceedings of the 7th international conference, EGOV 2008*, Springer, Heidelberg, Berlin, 1-12.
- Sorrentino, M., & Simonetta, M. (2013). Incentivising inter-municipal collaboration: the Lombard experience. *Journal of Management Governance*, 17, 887-906.
- Souza Neto, J. (2013). Metamodel of the IT Governance Framework COBIT. *Journal of Information Systems and Technology Management*, 10(3), 521-540.
- Teece, D. J., Pisano, G., & Shuen, A. (1997). Dynamic capabilities and strategic management. *Strategic Management Journal*, 18(7), 509-533.
- The Finnish Government (11th September 2015). Decision on next steps in reform package on healthcare, social welfare and autonomous regions. Retrieved from http://valtioneuvosto.fi/artikkeli/-/asset_publisher/ 10616/ hallitus-paatti-sote-uudistuksen-jatkosta-ja-itsehallintoalueista?_101_INSTANCE_ 3wyslLo1Z0ni_languageId=en_US
- TOGAF 9.2. (n.d.). The Open Group. Retrieved from http://www.opengroup.org/subjectareas/enterprise/togaf
- Trang, S. T.-N., Opitz, N., & Kolbe, L. (2013). IT governance in a network context: Literature review and agenda for research. *Proceedings of the Nineteenth Conference on Information Systems* (AMCIS), Chicago, Illinois, US.
- Turel, O., & Bart, C. (2014). Board-level IT governance and organizational performance. *European Journal of Information Systems*, 23(2), 223-239.
- Van Grembergen, W., De Haes, S., & Guldentops, E. (2004). Structures, processes and relational mechanisms for IT governance. In W. Van Grembergen (Ed.), *Strategies for information technology governance* (pp. 1-360). Hershey, Pennsylvania, USA: Idea Group Publishing.
- Van Grembergen, W., & De Haes, S. (2008). *Implementing Information Technology Governance: Models, Practices and Cases.* Hershey, PA, USA: Idea Group Global.
- Van Grembergen, W., & De Haes, S. (2012). A research journey into enterprise governance of IT, business/IT alignment and value creation. In W. Van Grembergen (Ed.), *Business strategy and applications in enterprise IT governance* (pp. 1-13). Hershey, PA, USA: IGI Global.
- Venkatesh, V., Brown, S. A., & Bala, H. (2013). Bridging the qualitative-quantitative divide Guidelines for conducting mixed methods research in information systems. MIS Quarterly, 37(1), 21-54.
- Volter (2018). The search engine: International articles. Retrieved from https://www.exlibrisgroup.com/products/primo-discovery-service/
- Walsham, G. (1995). Interpretive case studies in IS research: Nature and method. *European Journal of Information Systems*, 4(2), 74-81.
- Walsham, G. (2006). Doing interpretive research. European Journal of Information Systems, 15, 320-330
- Webster, J., & Watson, R. T. (2002). Analyzing the past to prepare for the future: Writing a literature review. *MIS Quarterly*, 26(2), 13-23.
- Weill, P., & Ross, J. W. (2004). IT governance: How top performers manage IT decision rights for superior results. Boston, MA: Harvard Business School Press.
- Weill, P., & Ross, J. W. (2004). *It governance on one page* (MIT Sloan Working Paper, No. 4517-04; CIS Research Working Paper No. 349). Retrieved from dx.doi.org/10.2139/ssrn.664612.
- Weill, P., & Woodham, R. (2002). *Don't just lead, govern: Implementing effective it governance* (MIT Sloan School of Management Working Paper, No. 4237-02). Retrieved from http://ssrn.com/abstract_id=317319.

- Wiengarten, F., Humphreys, P., Cao, G., & McHugh, M. (2013). Exploring the important role of organizational factors in IT business value: Taking a contingency perspective on the resourcebased view, *International Journal of Management Reviews*, 15(1), 30-46.
- Williamson, O. E. (1975). Markets and hierarchies: Analysis and antitrust implications A study in the economics of internal organization. New York, NY: The Free Press.
- Williamson, O. E. (1985). The economic institutions of capitalism. New York, NY: The Free Press.
- Zaheer, A., & Venkatraman, N. (1995). Relational governance as an inter-organizational strategy: An empirical test of the role of trust in economic exchange. *Strategic Management Journal*, 16(6), 373-392.
- Zaheer, A., Gözübüyük, R., & Milanov, H. (2010). It's the connections: The network perspective in interorganizational research. *The Academy of Management Perspectives*, 24(1), 62.
- Zarvić, N., Stolze, C., Boehm, M., & Thomas, O. (2012). Dependency-based IT governance practices in inter-organisational collaborations: A graph-driven elaboration. *International Journal of Information Management*, 32(6), 541-549.
- Zhang, P., Zhao, K., & Kumar, R. L. (2016). Impact of IT governance and IT capability on firm performance. *Information Systems Management*, 33(4), 357-373.
- Xiao, J., Xie, K., & Hu, Q. (2013). Inter-firm IT governance in power-imbalanced buyer-supplier dyads: Exploring how it works and why it lasts. *European Journal of Information Systems*, 22(5), 512-528
- Xue, Y., Liang, H., & Boulton, W. R. (2008). Information technology governance in information technology investment decision processes: The impact of investment characteristics, external environment and internal context. MIS Quarterly, 32(1), 67–96.
- Yin, R. K. (2009). Case study research: Design and methods (4th ed.). Thousand Oaks, CA: Sage Publications.

Appendix

Interview guidelines/questions/discussion themes in English

This guideline is a translated version of the guidelines for the interviewer. The target for the guidelines is to highlight the topics, which should be discussed during the interview. The guidelines are NOT meant to be followed meticulously but rather to provide discussion topics on a general level.

- A. Background to ICT cooperation
- 1. Why do you cooperate in the field of ICT with other municipalities?
- 2. Was this the original reason for ICT cooperation (if not, please specify)?
- 3. Who initially decided to launch ICT cooperation?
- 4. What were your initial expectations for ICT cooperation?
- 5. Which is the essential criterion for ICT cooperation: cost savings, or more efficient organization of work (choose another and justify)?
- B. Practice of cooperation
- 1. What does ICT cooperation entail in practice?
- 2. How is ICT cooperation organized?
- 3. How is the success of ICT cooperation evaluated?
- 4. Do you follow the COBIT model for IT governance, cooperation and ICT outsourcing?
- C. Impact of ICT cooperation
- 1. Who benefits from ICT cooperation?
 - i Within the municipality?
 - ii. Other (external) actor?

- 2. What are the costs of ICT cooperation (direct and indirect costs)?
- 3. What are the benefits of ICT cooperation?
 - i In terms of costs?
 - ii. In terms of work organization (short and long term)?
- 4. How has ICT cooperation worked so far, and what have been its real consequences?
- 5. Have there been any problems with ICT cooperation (if so, how have they been addressed)?
- D. The future of ICT cooperation
- 1. What do you expect from ICT cooperation in the future (within 1-3 and 4-5 years)?
- 2. How should ICT cooperation be improved in practice?
- 3. How could the IT governance structure be developed?
- E. Other comments



